Uttarakhand Tourism Development Board

GOVERNMENT OF UTTARAKHAND

Request for Proposal

For

Operation, Maintenance and Management of North

Corbett Resort at Kotdwar, Uttarakhand

Uttarakhand Tourism Development Board

Disclaimer

The information contained in this Request for Proposal ("RFP") Document or subsequently provided to Bidder, whether verbally or in documentary form by or on behalf of the Uttarakhand Tourism Development Board (UTDB), Government of Uttarakhand ("Government Representatives") or any of their employees or Uttarakhand Public Private Partnership Cell, is provided to the Bidder on the terms and conditions set out in this RFP Document and any other terms and conditions subject to which such information is provided.

This RFP Document is not an agreement and is not an offer or invitation by the Government Representatives to any party other than the Bidders who are qualified to submit the Proposal (Bidders). The purpose of this RFP Document is to provide the Bidder with information to assist the formulation of their proposals. This RFP Document does not purport to contain all the information each Bidder may require. This RFP Document may not be appropriate for all persons, and it is not possible for the Government Representatives, their employees or advisors, Uttarakhand Public Private Partnership Cell to consider the investment objectives, financial situation and particular needs of each party who reads or uses this RFP Document. Each Bidder should conduct its own investigations and analysis and should check the accuracy, reliability and completeness of the information in this RFP Document Representatives, their employees and advisors make no representation or warranty and shall incur no liability under any law, statute, rules or regulations as to the accuracy, reliability or completeness of the RFP Document.

The Government Representatives may in their absolute discretion, but without being under any obligation to do so, update, amend or supplement the information in this RFP Document. The Procurement Entity is also at liberty to call off this process, anytime during the bidding process, without giving any reason whatsoever.

Contents of RFP Document

Part I	Instructions to Bidders	Pages: 4 – 36
Part II	Draft Concession Agreement	Pages: 37 – 93

Part I Instructions to Bidders

Table of Contents

Sl. No	Contents	Page No.		
	Cover Page	1		
	Disclaimer	2		
	Content of RFP Document	3		
1	Instructions to Bidders	4		
2	Table of Contents	5		
3	Bidding Procedure	6		
4	Schedule of Bidding Process	16		
5	Scope of the Project	17		
6	Evaluation Methodology			
7	Fraud and Corrupt Practices	22		
	Appendices			
1	Format for Covering Letter cum Project Undertaking	23		
2	Format for Anti-Collusion Certificate	24		
3	Format for Performance Bank Guarantee	25		
4	Format for Technical Capability	27		
5	Format for Financial Proposal	28		
6	Format for Financial Capability	29		
7	Power of Attorney for signing of Application	30		
8	Format for Power of Attorney for Lead Member	31		
9	Format for Joint Bidding Agreement	33		

1 Bidding Procedure

A. General

- 1.1 Scope of Proposal
 - Tourism Development Board, Government of Uttarakhand, Uttarakhand (hereinafter referred to as "UTDB") invites detailed Proposals (Key Submissions and Financial Proposals) from Bidders who fulfil the prequalification criteria indicated in clause 1.22.2 for **Operation**, Maintenance and Management of North Corbett Resort at Kotdwar, Uttarakhand for a concession period of 30 (Thirty) years.
- 1.1.1 UTDB has developed a state of the art resort at north entry of Corbett National Park at Kotdwar and has 16, Cottages, 8 Deluxe Rooms, 20 Bed Dormatory and 1 banquet hall along with requisite infrastructure and now invites suitable public/private sector partners to manage the same.
- 1.1.2 The Proposals would be evaluated on the basis of the evaluation criteria set out in this RFP Document ("Evaluation Methodology") in order to identify the successful Bidder for the Project ("Successful Bidder"). The Successful Bidder would then have to enter into a Concession Agreement with UTDB and perform the obligations as stipulated therein, in respect of the Project.
- 1.1.3 Terms used in this RFP Document which have not been defined herein shall have the meaning ascribed thereto in the Draft Concession Agreement.

1.2 Eligible Bidders

- 1.2.1 Bidders who fulfil the pre-qualification criteria indicated in clause 1.22.2 are eligible to submit Technical and Financial Proposal in response to this RFP Document.
- 1.2.2 Bidders shall provide such evidence of their continued eligibility to the satisfaction of UTDB, as UTDB may reasonably request.
- 1.2.3 A Power of Attorney for signing of Proposals needs to be furnished in favour of the Authorized Signatory as per format provided in Appendix-7.
- 1.2.4 In the case of a Consortium, the Members should submit a Power of Attorney in favour of the Lead Member as per format provided in Appendix-8.
- 1.2.5 In case the Bidder is a Consortium, it shall, in addition to forming a Special Purpose Vehicle (SPV), comply with the following additional requirements:
 - a) Number of members in a consortium shall not exceed 3 (three),
 - b) Subject to the provisions of sub-clause (a) above, the Application should contain the information required for each member of the Consortium;
 - c) Members of the Consortium shall nominate one member as the lead member (the "Lead Member"), who shall have an equity share holding of at least 51% (Fifty-One per cent) of the paid up and subscribed equity of the SPV. The nomination(s) shall be supported by a Power of Attorney, as per the format at Appendix-8, signed by all the other members of the Consortium;
 - d) The Application should include a brief description of the roles and responsibilities of individual members, particularly with reference to financial, technical and O&M obligations;
 - e) An individual Bidder cannot at the same time be member of a Consortium. Further, a member of a particular Bidder Consortium cannot be member of any other Bidder Consortium;
 - f) The members of a Consortium shall form an appropriate SPV to execute the Project, if awarded to the Consortium;
 - g) Members of the Consortium shall enter into a binding Joint Bidding Agreement, substantially in the form specified at Appendix-9 (the "Jt. Bidding Agreement"), for the purpose of making the Application and submitting a Bid in the event of being short-listed.
 - h) The Jt. Bidding Agreement, to be submitted along with the Proposal, shall,

inter alia:

- i. Convey the intent to form an SPV with shareholding/ ownership equity commitment(s) in accordance with this RFP, which would enter into the Concession Agreement and subsequently perform all the obligations of the Concessionaire in terms of the Concession Agreement, in case the concession to undertake the Project is awarded to the Consortium;
- ii. Clearly outline the proposed roles and responsibilities, if any, of each member;
- iii. Commit the minimum equity stake to be held by each member;
- iv Commit that the Lead member, whose experience will be evaluated for the purposes of this RFP, shall subscribe to 51% (Fifty One per cent) or more of the paid up equity of the SPV and shall further commit that such member shall, for the entire concession period hold equity share capital not less than 51% (Fifty One per cent) of the paid up equity share capital of the SPV.
- v. Members of the Consortium undertake that they shall collectively hold 100% (One hundred per cent) of the paid-up equity of the SPV at all times until the end of the concession period.
- vi. Include a statement to the effect that all members of the Consortium shall be liable jointly and severally for all obligations of the Concessionaire in relation to the Project until the commencement of commercial operations of the Project is achieved in accordance with the Concession Agreement.
- vii. Except as provided under this RFP and the Bidding Documents, there shall not be any amendment to the Jt. Bidding Agreement without the prior written consent of UTDB.
- 1.2.6 A Bidder shall not have a conflict of interest (the "**Conflict of Interest**") that affects the Bidding Process. Any Bidder found to have a Conflict of Interest shall be disqualified. A Bidder shall be deemed to have a Conflict of Interest affecting the Bidding Process, if:
 - i. the Bidder, its Member or Associate (or any constituent thereof) and any other Bidder, its Member or any Associate thereof (or any constituent thereof) have common controlling shareholders or other ownership interest; provided that this disqualification shall not apply in cases where the direct or indirect shareholding of a Bidder, its Member or an Associate thereof (or any shareholder thereof having a shareholding of more than 5 per cent of the paid up and subscribed share capital of such Bidder, Member or Associate, as the case may be) in the other Bidder, its Member or Associate is less than 5 per cent of the subscribed and paid up equity share capital thereof; provided further that this disqualification shall not apply to any ownership by a bank, insurance company, pension fund or a public financial institution referred to in section 2(72) of the Companies Act 2013. For the purposes of this Clause 1.2.6 (i), indirect shareholding held through one or more intermediate persons shall be computed as follows:
 - a. where any intermediary is controlled by a person through management control or otherwise, the entire shareholding held by such controlled intermediary in any other person (the "**Subject Person**") shall be taken into account for computing the shareholding of such controlling person in the Subject Person; and
 - b. subject always to sub-clause (a) above, where a person does not exercise control over an intermediary, which has shareholding in the Subject Person, the computation of indirect shareholding of such person in the Subject Person shall be undertaken on a proportionate basis; provided, however, that no such shareholding shall be reckoned under this sub-clause (b) if the shareholding of such person in the intermediary is less than 26% of the subscribed

UTDB

and paid up equity shareholding of such intermediary;

or

ii. a constituent of such Bidder is also a constituent of another Bidder; or

- iii. such Bidder, or any Associate thereof receives or has received any direct or indirect subsidy, grant, concessional loan or subordinated debt from any other Bidder, or any Associate thereof or has provided any such subsidy, grant, concessional loan or subordinated debt to any other Bidder, its Member or any Associate thereof;
- or
 - iv. such Bidder has the same legal representative for purposes of this Application as any other Bidder;
- or
- v. such Bidder, or any Associate thereof has a relationship with another Bidder, or any Associate thereof, directly or through common third party/ parties, that puts either or both of them in a position to have access to each other's information about, or to influence the Application of either or each other;
- or

1.2.7

- vi. Such Bidder or any Associate thereof has participated as a consultant to UTDB in the preparation of any documents, design or technical specifications of the Project.
- A Bidder shall be liable for disqualification if any legal, financial or technical adviser of UTDB in relation to the Project is engaged by the Bidder, its Member or any Associate thereof, as the case may be, in any manner for matters related to or incidental to the Project. For the avoidance of doubt, this disqualification shall not apply where such adviser was engaged by the Bidder, its Member or Associate in the past but its assignment expired or was terminated 6 (six) months prior to the date of issue of this RFP. Nor will this disqualification apply where such adviser is engaged after a period of 3 (three) years from the date of commercial operation of the Project.

Explanation: In case a Bidder is a Consortium, then the term Bidder as used in this Clause 1.2, shall include each Member of such Consortium.

For purpose of this RFP, Associate means, in relation to the Bidder/ Consortium Member, a person who controls, is controlled by, or is under the common control with such Bidder/ Consortium Member (the "Associate"). As used in this definition, the expression "control" means, with respect to a person which is a company or corporation, the ownership, directly or indirectly, of more than 50% (fifty percent) of the voting share of such person, and with respect to a person which is not a company or corporation, the power to direct the management and policies of such person by operation of law.

1.3 Additional Requirements for Proposals Submitted by a Consortium

(a) Wherever required, the Proposal shall contain the information required for each of the member of the Consortium;

(b) The Proposal shall be signed by the duly Authorised Signatory of the Lead Member and shall be legally binding on all the members of the Consortium;

(c) All witnesses and sureties shall be persons of status and probity and their full names and addresses shall be stated below their signature. All signatures in the Proposal documents shall be dated.

1.4 Change in Composition of the Bidder

- 1.4.1 Any change in composition of the Bidder after the Pre-Qualification stage would be at the sole discretion of UTDB and with prior written permission of UTDB.
- 1.4.2 Each Bidder shall submit only one (1) Proposal for the project in response to this

RFP Document. Any entity, which submits or participates in more than one Proposal will be disqualified and will also cause the disqualification of Consortium in which it is a member.

1.5 Proposal Preparation Cost

1.5.1 The Bidder shall be responsible for all the costs associated with the preparation of its Proposal and its participation in the bidding process. UTDB will not be responsible or in any way liable for such costs, regardless of the conduct or outcome of bidding.

1.6 Contents of RFP Document

1.6.1 RFP Document consists of 2 Parts as listed below and would include any addenda issued in accordance with Clause 1.8:-

Issued in accordance with Clause 1.6		
Part I	Instructions to Bidders	
Part II	Draft Concession Agreement	

1.7 Clarifications

1.7.1 Bidders requiring any clarification on the RFP Document may notify UTDB in writing or by facsimile within such date as specified in the Schedule of Bidding Process. Based on its sole discretion, UTDB may forward to all Bidders, copies of UTDB response, including a description of the enquiry but without identifying its source.

1.8 Amendment of RFP Document

- 1.8.1 At any time prior to the Proposal Due Date, UTDB may, for any reason, whether at its own initiative or in response to clarifications requested by a Bidder, modify the RFP Document by the issuance of Addenda.
- 1.8.2 Any Addendum / Corrigendum thus issued will be posted on the website: <u>www.uktenders.gov.in</u>.
- 1.8.3 In order to afford Bidders reasonable time in which to take an Addendum into account, or for any other reason, UTDB may, at its own discretion, extend the Proposal Due Date.

B. Preparation and Submission of Proposal

1.9 Language and Currency

- 1.9.1 The Proposal and all related correspondence and documents shall be written in the English language. Supporting documents and printed literature furnished by the Bidder with the Proposal may be in any other language provided that they are accompanied by an appropriate translation into English. Supporting materials that are not translated into English may not be considered. For the purpose of interpretation and evaluation of the Proposal, the English language translation shall prevail.
- 1.9.2 The currency for the purpose of the Proposal shall be the Indian Rupee (INR).

1.10 Bid Security

1.10.1

- (i) Bid Security of Rs. 16.60 Lakh (Rupees Sixteen Lakh Sixty Thousand) only online
- (ii) Proposals not accompanied by Bid Security shall be summarily rejected.
- (iii) The Proposal fee and Bid Security should be remitted through net banking. The details are:

Beneficiary name – CEO Uttarakhand Tourism Development Board Dehradun

Bank's name – Allahabad Bank

Account Number - 50517934250

Branch – Nimbuwala, Garhi Cantt

Address – Nimbuwala, Garhi Cantt, Dehradun IFSC – ALLA0213398 GST No. – 05AAALU0031F1ZK PAN – AAALU0031F

- 1.10.2 The Bid Security shall be kept valid through the Proposal Validity Period and would be required to be extended if so required by UTDB.
- 1.10.3 The Bid Security shall be returned to unsuccessful Bidders within a period of thirty (30) days from the date of announcement of the Successful Bidder. The Bid Security submitted by the Successful Bidder shall be released upon furnishing of the Performance Security in the form and manner stipulated in the Draft Concession Agreement.
- 1.10.4 The Successful Bidder's Bid Security will be returned, without any interest, upon the signing of the Concession Agreement and furnishing the Performance Security in accordance with the provisions thereof.
- 1.10.5 The Bid Security shall be forfeited in the following cases:
 - (a) If the Bidder withdraws its Proposal except as provided in Clause 1.18.1;

(b) If the Bidder withdraws its Proposal during the interval between the Proposal Due Date and expiration of the Proposal Validity Period; and

(c) If the Successful Bidder fails to provide the Performance Security within the stipulated time or any extension thereof provided by UTDB.

1.10.6 The bidders (enterprises) complying with the provisions of Uttarakhand Government Order No. 1542/VII-3-19/143-Industry/2003 dated 20th August 2019 are exempted from payment of Tender Fee and Earnest Money Deposit (EMD).

1.11 Validity of Proposal

- 1.11.1 The Proposal shall indicate that it would remain valid for a period not less than 120 (One Hundred and Twenty) days from the Proposal Due Date (Financial Proposal Validity Period). UTDB reserves the right to reject any Proposal that does not meet this requirement.
- 1.11.2 Prior to expiry of the original Proposal Validity Period, UTDB may request that the Bidders extend the period of validity for a specified additional period. A Bidder may refuse the request without forfeiting its Bid Security. The bid of any bidder refusing to extend the bid security shall be returned and shall not be included in the further bid process. A Bidder agreeing to the request of extending the bid security will not be allowed to modify its Proposal, but would be required to extend the validity of its Bid Security for the period of extension and comply with Clause 1.10 of this document in all respects.
- 1.11.3 The Successful Bidder shall extend the Proposal Validity Period till the date of execution of the Concession Agreement.

1.12 Bidder's Responsibility

- 1.12.1 The Bidder is expected to examine carefully the contents of all the documents provided. Failure to comply with the requirements of RFP Document will be at the Bidder's own risk.
- 1.12.2 It would be deemed that prior to the submission of Proposal, the Bidder has:
 - (a) made a complete and careful examination of requirements, and other information set forth in this RFP Document;
 - (b) received all such relevant information as it has requested from UTDB; and
 - (c) made a complete and careful examination of the various aspects of the Project including but not limited to:
 - (i) the conditions of the access roads and utilities in the vicinity of the Project Site
 - (ii) conditions affecting transportation, access, disposal, handling and storage of the materials

- (iii) clearances obtained by UTDB for the Project; and
- (iv) All other matters that might affect the Bidder's performance under the terms of this RFP Document.
- 1.12.3 UTDB shall not be liable for any mistake or error or neglect by the Bidder in respect of the above.

1.13 Pre-Proposal Meeting

- 1.13.1 To clarify and discuss issues with respect to the Project and the RFP Document, UTDB may hold Pre-Proposal meeting/s on date specified in the Schedule of bidding process.
- 1.13.2 Prior to the Pre-Proposal meeting/s, the Bidders may submit a list of queries and propose deviations, if any, to the Project requirements and/or the Draft Concession Agreement. UTDB may amend the RFP Document based on inputs provided by Bidders that may be considered acceptable in its sole discretion.
- 1.13.3 Bidders may note that UTDB will not entertain any deviations to the RFP Document at the time of submission of the Proposal or thereafter. The Proposal to be submitted by the Bidders will be unconditional and unqualified and the Bidders would be deemed to have accepted the terms and conditions of the RFP Document with all its contents including the Draft Concession Agreement. Any conditional Proposal shall be regarded as non-responsive and would be liable for rejection.
- 1.13.4 UTDB will endeavor to hold the meeting as per Schedule of Bidding Process. The details of the meeting will be separately communicated to the Bidders.
- 1.13.5 Attendance of the Bidders at the Pre-Proposal meeting is not mandatory. However, subsequent to the meeting, UTDB may not respond to queries from any Bidder who has not attended the Pre-Proposal meeting.
- 1.13.6 All correspondence / enquiries should be submitted only through e-mail PPP Cell pppcell.utdb@gmail.com.
- 1.13.7 No interpretation, revision, or other communication from UTDB regarding this solicitation is valid unless posted on website: <u>www.uktenders.gov.in</u>.

1.14 Format and Signing of Proposal

- 1.14.1 Bidders would provide all the information as per this RFP Document and in the specified formats. UTDB reserves the right to reject any Proposal that is not in the specified formats.
- 1.14.2The Proposal should be submitted in 3 parts:
Part 1: Key Submissions, which would include:
 - i. Covering Letter cum Project Undertaking as per Appendix 1 stating the Proposal Validity Period
 - ii. Anti-Collusion Certificate as per Appendix 2
 - iii. Bid Security as per clause 1.10
 - iv. Power of Attorney for Signing of Application as per Appendix 7
 - v. Power of Attorney for Lead Member of Consortium as per Appendix 8
 - vi. Bank Draft towards cost of RFP Document

Part 2: Technical Proposal as per format set out in Appendix - 4 (*Not Required*). **Part 3: Financial Proposal** as per the format set out in Appendix -5.

1.14.3 The Bidder shall prepare and submit online through website: <u>www.uktenders.gov.in</u> scanned copies of original documents comprising the Proposal as described in Clause 1.14.

1.15 Submission of Proposals

- 1.15.1 The Bidder shall submit / upload Technical Proposal and the Financial Proposal as per provisions on the e-portal <u>www.uktenders.gov.in</u>.
- 1.15.2 Originals of (a) Proposal fee, (b) Bid Security, (c) Power of Attorney and (d) JV Agreement (if applicable) are to be submitted to Chief Executive Officer,

Uttarakhand Tourism Development Board (UTDB) Pt. Deen Dayal Upadhyay, Paryatan Bhawan, Near O.N.G.C. Helipad, Nimbuwala, Garhi Cantt, Dehradun, **before Technical Proposal opening.** The proposal inviting authority shall not be held liable for any delays in the receipt of these documents. <u>No other original document need to be submitted</u>.

1.16 Proposal Due Date

- 1.16.1 Proposals should be submitted / uploaded on the website: <u>www.uktenders.gov.in</u>. Submission of proposals through any other mode is not acceptable and shall be rejected.
- 1.16.2 UTDB, at its sole discretion, may extend the Proposal Due Date by issuing an Addendum in accordance with Clause 1.16.1

1.17 Late Proposals

1.17.1 The portal will not accept any proposal after the specified due time as per its server clock.

1.18 Modification and Withdrawal of Proposals

- 1.18.1 The Bidder may modify or withdraw its Proposal on e-portal before the Proposal Due Date and time. No Proposal shall be modified or withdrawn by the Bidder after the Proposal Due Date and time.
- 1.18.2 Withdrawal of a Proposal during the interval between the Proposal Due Date and expiration of the Proposal Validity Period would result in forfeiture of the Bid Security in accordance with Clause 1.10.5 of this document.

1.19 Tests of responsiveness

- 1.19.1 Prior to evaluation of Proposals, UTDB will determine whether each Proposal is responsive to the requirements of the RFP Document. A Proposal shall be considered responsive if
 - (a) it is received only on e-portal.
 - (b) Submitted as per provisions of the e-portal.
 - (c) it contains the information and documents as requested in the RFP Document.
 - (d) it contains information in formats specified in the RFP Document.
 - (e) it mentions the proposal validity period as set out in Clause 1.11.
 - (f) it provides the information in reasonable detail. ("Reasonable Detail" means that, but for minor deviations, the information can be reviewed and evaluated by UTDB without communication with the Bidder). UTDB reserves the right to determine whether the information has been provided in reasonable detail.
 - (g) there are no inconsistencies between the Proposal and the supporting documents.
- 1.19.2 A Proposal that is substantially responsive is one that conforms to the preceding requirements without material deviation or reservation. A material deviation or reservation is one
 - (a) which affects in any substantial way, the scope, quality, or performance of the Project,
 or
 - (b) which limits in any substantial way, inconsistent with the RFP Document, UTDB's rights or the Bidder's obligations under the Concession Agreement, or
 - (c) which would affect unfairly the competitive position of other Bidders presenting substantially responsive bids.
- 1.19.3 UTDB reserves the right to reject any Proposal which in its opinion is nonresponsive and no request for modification or withdrawal shall be entertained by UTDB in respect of such Proposals.
- 1.19.4 Conditional proposal shall not be considered. Any bid found to contain conditions

attached, will be rejected.

1.20 Confidentiality

1.20.1 Information relating to the examination, clarification, evaluation and recommendation shall not be disclosed to any person not officially concerned with the process. UTDB will treat all information submitted as part of Proposal in confidence and will ensure that all who have access to such material treat it in confidence. UTDB will not divulge any such information unless it is ordered to do so by any Government authority that has the power under law to require its disclosure.

1.21 Clarifications

- 1.21.1 To assist in the process of evaluation of proposals, UTDB may, at its sole discretion, ask any Bidder for clarification including additional information and documents. In case of any additional documents, same can be accepted, only if they are of historical nature i.e., either the documents or facts in the documents should have existed prior to be proposal submission time and same could be verified independently. However, no change in the substance of the proposal would be permitted by way of such clarifications. The request for clarification and the response shall be in writing or e-mail or by facsimile.
- 1.21.2 UTDB reserves the right to independently verify by a team of Officers of UTDB or independently facts and figures provided in the documents submitted by the Bidders; in addition to right to disqualify, reject and/ or accept any and all submissions or other information and/ or evidence submitted by or on behalf of any Bidder.

1.22 Proposal Evaluation

- 1.22.1 To assist in the examination, evaluation, and comparison of Proposals, UTDB may utilise the services of consultant/s or advisor/s.
- 1.22.2 Before opening and evaluation of the Technical Proposals, Bidders are expected to meet the following pre-qualification criteria which would be a part of the Technical Proposal. Bidders failing to meet these criteria or not submitting requisite proof for supporting pre-qualification criteria are liable to be rejected at the Technical Proposal level.

Sl.	Criteria	Reference Details
No		
1	The bidder should be a	Certificate of registration or
	public/ private organization registered	incorporation/ Any other
	in India as a Proprietor/Company/	relevant document
	society/firm under respective Act	
2	The bidder should have financial	CA certificate
	capability as per criteria given under	
	Evaluation Methodology (Section 4)	
3	The bidder should have technical	CA certificate or self-
	capability as per criteria given under	declaration
	Evaluation Methodology (Section 4)	

- 1.22.3 The bidders shall submit Audited balance sheet along with all schedules as a proof to establish non participation of associate companies in the bid process.
- 1.22.4 The technical and financial capability of all the members of consortium shall be considered for the purpose of evaluation.
- 1.22.5 However, experience gained from development, operation and management of the same project by more than one member of the Consortium shall be counted as experience of only one consortium member.
- 1.22.6 The bidder shall attach signed agreement between consortium members specifying the equity participation in the project.

- 1.22.7 Evaluation of Proposals will be done in two stages.
- 1.22.8 In Stage I of Proposal Evaluation, the Key Submissions submitted by the Bidders shall be checked for responsiveness with the requirements of the RFP Document.
- 1.22.9 The evaluation of Technical Proposal of a Bidder shall be taken up only after the contents of the Key Submissions are found to meet the requirements of this RFP Document. UTDB reserves the right to reject the Proposal of a Bidder without opening the Technical Proposal if the contents of Key Submissions are not substantially responsive with the requirements of this RFP Document.
- 1.22.10 After checking the responsiveness of the Key submission, the Technical Proposal will be opened. The bidder should meet the minimum requirement given under **Evaluation Methodology** (Section 4)

1.23 Evaluation of the Financial Proposal

- 1.23.1 In Stage II, the Financial Proposals of all the Bidders who pass the Stage I evaluation will be opened in the presence of the Bidders' representatives who choose to attend. The Bidders' representatives who choose to be present shall be required to sign and record their attendance.
- 1.23.2 Evaluation of Financial Proposal shall be done as per section 4.2 of this RFP Document.

1.24 Notifications

- 1.24.1 Upon acceptance of the Financial Proposal of the Preferred Bidder with or without negotiations, UTDB shall declare the Preferred Bidder as Successful Bidder.
- 1.24.2 UTDB will notify the Successful Bidder by facsimile or e-mail and by a letter (Speed Post / Registered Post) that its Proposal has been accepted.

1.25 UTDB's Right to Accept or Reject Proposal

- 1.25.1 UTDB reserves the right to accept or reject any or all of the Proposals without assigning any reason and to take any measure as it may deem fit, including annulment of the bidding process, at any time prior to award of Project, without liability or any obligation for such acceptance, rejection or annulment.
- 1.25.2 UTDB reserves the right to reject any Proposal including that of the Preferred bidder if:

(a) at any time, a material misrepresentation is made or uncovered; If a fraud or fraudulent practice adopted by any bidder is established, the bidder may be blacklisted and /or appropriate legal proceedings may be initiated against such bidder as per the prevailing laws OR

(b) the Bidder does not respond promptly and thoroughly to requests for supplemental information required for the evaluation of the Proposal.

This would lead to the disqualification of the Bidder. If the Bidder is a Consortium, then the entire Consortium would be disqualified/ rejected. If such disqualification/ rejection occur after the Financial Proposals have been opened and the highest ranked Bidder gets disqualified / rejected, then UTDB reserves the right to:

(a) either invite the next highest ranked Bidder to match the Financial Proposal submitted by the highest ranked Bidder; OR

(b) take any such measure as may be deemed fit in the sole discretion of UTDB, including annulment of the bidding process.

1.26 Acknowledgement of Letter of Acceptance (LOA)

1.26.1 Within two (2) weeks from the date of issue of the LOA, the Successful Bidder shall acknowledge the receipt of LOA.

1.27 Execution of Concession Agreement

1.27.1 The Successful Bidder shall execute the Concession Agreement within 30 (Thirty)

days of the issue of LOA or such time as indicated by UTDB.

1.27.2 UTDB will promptly notify other Bidders that their Proposals have been unsuccessful and their Bid Security will be returned as promptly as possible in any case not later than 30 (thirty) days from the date of announcement of the Successful Bidder.

1.28 Performance Security

- 1.28.1 The Successful Bidder shall furnish Performance Security by way of an irrevocable and unconditional Bank Guarantee issued by a scheduled bank located in India in favour of UTDB as required under the Concession Agreement.
- 1.28.2 The Performance guarantee shall be forfeited and en-cashed in the following cases:(a) If the Successful Bidder withdraws from the project midway during the project term

(b) Any other act or acts of the successful bidder which renders the project unoperational and UTDB establishes sufficient reasons to forfeit the performance guarantee.

(c) If the bidder fails to fulfil its financial obligation to UTDB regarding payment of Annual Concession Fee or any other payables.

(d) If the bidder's action/negligence damages the project asset of UTDB.

(e) If the bidder fails to meet the service levels as agreed.

1.29 Cost of RFP document:

1.29.1 The RFP document can be downloaded from www.uktenders.gov.in. The cost of RFP document is Rupees 10,000/- + Applicable GST to be paid online as per details given under ITB Clause No. 1.10.1. The Proposal Document Fee is nonrefundable. Non-submission of Proposal Document Fee along with the technical proposal shall be summarily rejected. The bidders (enterprises) complying with the provisions of Uttarakhand Government Order No. 1542/VII-3-19/143-Industry/2003 dated 20th August 2019 are exempted from payment of Tender Fee and Earnest Money Deposit (EMD).

2. Schedule of Bidding Process

2.1

UTDB would endeavor to adhere to the following schedule:

Sl. No.	Activity Description	Time	Date
1	Pre-proposal meeting	15:00 Hrs	18/03/2021
2	Last date for receiving queries	15:00 Hrs	18/03/2021
3	Proposal submission start date		11/03/2021
4	Proposal Due Date	15:30 Hrs	03/04/2021
5	Opening of Technical Proposals 16:00 Hrs 03/04/2		03/04/2021
6	Opening of Financial Proposal	To be	e intimated

Pre-proposal meeting and Opening of Proposals: Venue:

Due to prevailing Covid-19 pandemic, Pre-proposal meeting shall be held online as per schedule.

Interested bidders willing to participate in Pre-proposal meeting are requested to share their intent of joining the meeting online on ppcell.utdb@gmail.com mentioning "Pre-Proposal Meeting – Kotdwar Resort", so that meeting link can be forwarded to them.

3. Scope of the Project

3.1 Introduction:

a) Government of Uttarakhand through Tourism Department is developing a state of the art North Corbett Resort on 11 acre land along with visitor facilitation /information and an interpretation centre at North Corbett entry point at Kotdwar under the ADB funded Infrastructure Development Investment Program for Tourism (IDIPT). The project facility incudes:-

- Visitor facilitation/ Information centre and an Interpretation centre
- Convention centre
- Cottage 16 Numbers
- Executive Bedroom 4 Numbers
- Deluxe Bedroom 4 Numbers
- Overhead water storage tank
- Electrical room
- Staff and driver dormitory
- Manager room
- Parking area
- Musical & Dancing Fountain
- Food court
- Generator shed
- Artificial Climbing Wall

b) Understanding for that has been arrived between UTDB and the Government providing authority to operate, maintain & manage these facilities through selected public/private partner.

c) UTDB proposes to handover the North Corbett Resort/facility to selected bidder for Operation, Maintenance & Management on PPP basis.

d) The Concession Period shall be Thirty (30) years. The Agreement shall not be renewable. However the present Operator shall be eligible to bid at the next bidding, there will however be no First Right of Refusal available to the present Operator. **Project Components**

Project Components	
a) The following are the details of the North Corbett Resort/facility:	

Sl.No.	Key Information	Details
1.	Project Title	Operation, Maintenance and Management of North Corbett Resort, Kotdwar, Uttarakhand on PPP Basis.
2.	Location	North Corbett Entry, Kotdwar, Uttarakhand, India
3.	Approximate Project Area	Plan Area of Site: 11 Acre
4.	Minimum Obligations	The selected Public/Private operator is required to operate, maintain and manage the project and related services as eco resort as mentioned in the draft Concession Agreement including procurement of the associated martial, equipment, articles and necessary inventories from the reputed vendors / manufacturers as approved by the Authority. The successful bidder can add rooms/ cottage not more than 16 in existing facility with prior approval of the Authority.
5.	O & M Period	30 (Thirty) years from the Appointed Date as

			the terms of the Concession Agreement.	
6.	Statutory	Clearance	To be obtained by the Selected Public/Private	
	and	Necessary	Operator. The Authority shall provide assistance	
	Licenses	-	to the Selected Public/Private Operator in	
			procurement of the said clearances/licenses.	

b) The already built facility should not be tampered with or any fresh construction by the successful bidder.

c) **Approach, Water & Power**: The facility already has these facilities. Public/private partner can augment the capacity by investing from his own end, with prior approval of UTDB.

d) The facility shall be handed over to the selected public/private partner as is where is basis and shall be taken back after 30 years of concession period on same conditions of as is where is basis.

e) The successful public/private partner shall not use the facility for any unlawful activities.

f) Any complaint received for such unlawful activity shall be investigated by the appropriate authority and if found correct then UTDB reserves the right to terminate the Concession Agreement as per the Concession Agreement. UTDB reserves the right to claim damages from the private party as it may be advised under applicable laws.

Broad Scope of Work:

The scope of work for the Public/Private Partner (Operator) will include:

- a) Operation, Maintenance and Management of North Corbett Resort, Kotdwar, Uttarakhand as eco resort as per agreed terms throughout the concession period.
- b) May add rooms/ cottage in existing facility with prior approval of the Authority.
- c) Recruitment, management and training of staff required for the proposed project.
- d) Revenue Collection selected operator is free to fix and revise tariff.
- e) Manage all clearances and approvals for the Project, if required. UTDB will facilitate in managing clearances and approvals.
- f) Management of the entire facility and related services
- g) Avoid unauthorized usage of project infrastructure
- h) Timely payment of Concession Fee to UTDB.

3.2 Terms of Reference for the Public/Private Partner (Operator):

[This section aims to give as much as information as it possible in the document may not be complete in all respects. The prospective bidders are encouraged to visit site and satisfy themselves completely before submitting the bid. It will be assumed that any bidder submitting the bid will be aware of the facilities at the site.]

- 3.2.1 The broad scope of work for selected Operator shall include the following but not limited to:-
- 3.2.1.1 **Operation, Maintenance and Management:** The Selected Operator is required to undertake operations, maintenance and management of the Tourism Infrastructure and related facilities being developed in and around North Corbett Resort at Kotdwar to world class standards. The UTDB will transfer "Right to Use" to Operator for these assets and infrastructure for the entire Contract period. The Operator shall submit detailed Operation, Maintenance and Management Plan, Resource Mobilization Plan to UTDB for approval.

The responsibilities of the operator would be as follows:-

• <u>Operation and Management</u>: The Operator would be required to discharge its obligations as detailed out. The manpower of required skills shall be arranged by Operator for the Contract period at his own cost. The cost of utilities such as electricity, water, telephone and other operating costs will be borne by the Operator. All periodic upkeep, maintenance and repairs of the assets and facilities

transferred/developed under the contract will be carried out by the Operator. The inventory of Assets including all equipment and other infrastructure shall be provided by UTDB to selected bidder

- <u>Facility Management Mechanized Housekeeping</u>: Providing facility management service at the project area (Buildings and open premises including the parking area) by Sweeping, Scrubbing, Buffing, Vacuum Cleaning, Pressure Washing, etc. as per frequency of cleaning chart by using various machines, including using chemicals, toiletries and consumables etc. as per general conditions, additional conditionals and special conditions etc. The Operator shall be responsible for
 - Full range of cleaning/housekeeping activities;
 - Kitchen and restaurant;
 - Food and beverages service;
 - Minor maintenance service;
 - Provision of adequate supplies of linen and toiletries and other consumables;
 - Operation of facilities inclusive of the business centre, customer service centre, etc.;
 - Potable Water Supply;
 - Seepage Management; and
 - Pest Control & Garbage Management.
- <u>Landscaping & Garden Maintenance</u>: Landscaping and maintenance of garden/green areas including cleaning and sweeping, removal of weeds, application of soil, manure and chemical fertilizers, cutting of hedges, shrubs, trees, moving of lawn mower, daily watering to total garden etc.
- <u>Customer Service Help Desk</u>: The Operator shall Establish and maintain required level of Customer Service Helpdesk related to inquiries related to following areas. This will be measured objectively based on the customer feedback received a per the criteria laid down in the Key Performance Metrics in the Agreement
 - Tour Booking Information
 - Booking Information
 - Tourism Destination related queries & information
 - All the information related to Uttarakhand Tourism

The Help Desk would be operational all seven days. The persons attending Help Desk should be well conversant with English and Hindi.

- <u>Prohibitions:</u> Smoking, consumption of liquor, and chewing of tobacco etc. by the staff of the Operator is totally banned within the facility area of Adventure Tourism Centre
- 3.2.1.2 **Revenue Generation**: The Operator will be responsible for generation of revenues through the following revenue streams. The tariffs/rates would be decided by the Operator based on market conditions and tourist footfalls.
 - <u>Levy of Tariffs for Accommodation</u>: The Operator will be required levy tariffs for accommodation/stay facilities.
 - <u>Revenue from leasing of Cafeteria/kiosks</u>: The Operator would have option to lease out designated areas for cafeteria and kiosks. The lease rentals will be as per the current market rentals and be mutually decided between the lesser and lessees. The operator shall indemnify UTDB while subleasing, so that the sublease shall come to an end automatically when the lease period of the selected Operator comes to an end. In any case the lease period of any sublease shall not be more than the period of the lease given to the selected Operator.
 - <u>Revenue from Tour Operations</u>: The Operator will be allowed to conduct tour to tourist places through UTDB approved/registered tour operators.
 - <u>Conferences, Events & Exhibitions</u>: The Operator will be allowed to hold private conferences, events and exhibition. The events and exhibitions will be related to cultural and tourism related and will have prior approval from UTDB. The dates will

be kept in such a way that tourist footfalls for other festivals/events organized by UTDB / GMVN / KMVN are not affected.

- <u>Branding</u>: The Operator will have the rights to display their own corporate logos and corporate advertisements at pre-designated locations approved by UTDB. The Operator will also be required to accommodate UTDB's or GoUK's display materials in pre-designated locations as directed by UTDB.
- 3.2.1.3 **Environmental Management**: The entire O&M activities have to be carried out in such a way that it has least adverse impact on the environment. The periodic monitoring environmental conditions of the ambient air quality, noise level, water quality, in and around the project facilities as per prescribed norms of statutory authorities, shall be got done through an approved monitoring agency. The Operator, at its own cost, will take corrective actions, if any, as per observation of monitoring agency.
- 3.2.1.4 **Brand Building and Marketing**: The Operator shall undertake the required brand building, advertising and marketing activities across multiple media channels in order to facilitate visitors/tourists footfall and high occupancy rates and optimal use of facilities and services.

3.2.1.5 **Key Performance Indicators:**

The successful concessionaire, in consultation with UTDB, will develop Key Performance Indicators (KPIs) related to Customer Service and Branding of UTDB/Facilities under the project. UTDB will review the KPIs on regular basis and the concessionaire will take necessary measures for improvement of performance.

3.2.2 Other Conditions:

- There will always be ownership of UTDB on whole premises including land, infrastructures, open spaces etc. The successful bidder (the "Operator") shall have the license to use infrastructures only. Open spaces (landscaped area, pathways) will be common area that will be freely accessible for tourists any time, day. If developer wants to use these open spaces for specific purpose like cultural events/exhibition etc, prior approval has to be taken from competent authority of UTDB. UTDB has always it's right reserved to create temporary/permanent structures on open space as and when required and these types of temporary/permanent structures will not be a part of original agreement. However additional recreational facilities which are temporary in nature and in concurrence with the applicable Laws and Regulations may be created by operator in license period only and only by prior approval of competent authority not below Secretary, Tourism, Government of Uttarakhand.
- The Operator shall adhere to the national and state labor laws and shall be responsible for all liabilities arising due to any disputes thereof. UTDB shall not be held responsible for any labor disputes.

4 Evaluation Methodology

4.1 Evaluation of Technical Proposal

The Technical Capability Proposal for the Project would be evaluated on the following basis:

- **1.** For any Applicant to meet the experience criterion of qualification, they need to satisfy the following:
 - (a) Must have set up / managed at least 1 resort/ hotel having minimum of 24 rooms for at least 1 year during last five years prior to the last date for bid submission.
 - (b) Financial Capability of the applicant will be evaluated on the basis of the following:
 - Positive Net worth of minimum Rs. 207.55 Lakhs as on 31.03.2020
- Or
- 2. Positive Net worth of minimum Rs. 5.00 Crore as on 31.03.2020, however, such Bidder (s) will be required to have a tie-up for the entire duration of concession period to operate the facility, later, with any International / National Hotel operator chain which is running at least one property of 3-star category or above within India or Abroad, within 2 (two) months of signing of Concession Agreement

<u>Note</u>: Above qualifications should be met by at least one member in case of a Consortium.

4.1.1 The Financial bids of those bidders would be opened who meet the requirement given above.

4.2 Evaluation of Financial Proposals

- 4.2.1 The bidder shall quote a fixed Annual Concession Fee amount for first year. The minimum amount of Annual Concession Fee for the first year is fixed at Rs. 15.00 Lakh. The bidder shall quote Annual Concession Fee amount for the first year, which should not less than minimum amount fixed for Annual Concession Fee. more than this minimum amount. This Concession Fee shall increase from second year by 5% every year over the previous year Concession Fee commencing and payable from Second year.
- 4.2.2 The bidder quoting the highest amount of Annual Concession Fee for the 1st year shall be declared as the Successful Bidder for the facility.
- 4.2.3 In the event that two or more Bidders quote same amount, the UTDB may (In this order):
 - (a) invite fresh financial proposals from such concerned Bidders only; or
 - (b) take any such measure as may be deemed fit in its sole discretion including annulment of the bidding process.
- 4.2.4 UTDB may either choose to accept the Proposal of the Preferred Bidder or invite him for negotiations.
- 4.2.5 Upon acceptance of the Financial Proposal of the Preferred Bidders with or without negotiations, UTDB shall declare the Preferred Bidder as Successful Bidder.

5. Fraud and Corrupt Practices***

- 5.1 The Bidders and their respective officers, employees, agents and advisers shall observe the highest standard of ethics during the Bidding Process. Notwithstanding anything to the contrary contained herein, UTDB may reject an Application without being liable in any manner whatsoever to the Bidders if it determines that the Bidders has, directly or indirectly or through an agent, engaged in corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice in the Bidding Process.
- 5.2 Without prejudice to the rights of UTDB, if an Bidders is found by UTDB to have directly or indirectly or through an agent, engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice during the Bidding Process, such Bidders shall not be eligible to participate in any tender or RFP issued by UTDB or by any other agency of Government of Uttarakhand during a period of 2 (two) years from the date such Bidders are found by UTDB to have directly or indirectly or through an agent, engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice, as the case may be.
 - For the purposes of this Clause 5, the following terms shall have the meaning hereinafter respectively assigned to them:

(a) "corrupt practice" means (i) the offering, giving, receiving, or soliciting, directly or indirectly, of anything of value to influence the actions of any person connected with the Bidding Process (for avoidance of doubt, offering of employment to, or employing, or engaging in any manner whatsoever, directly or indirectly, any official of UTDB who is or has been associated in any manner, directly or indirectly, with the Bidding Process or the LOA or has dealt with matters concerning the Concession Agreement or arising therefrom, before or after the execution thereof, at any time prior to the expiry of one year from the date such official resigns or retires from or otherwise ceases to be in the service of UTDB, shall be deemed to constitute influencing the actions of a person connected with the Bidding Process); or (ii) save and except as permitted under sub clause (d) of Clause 1.2.5, engaging in any manner whatsoever, whether during the Bidding Process or after the issue of the LOA or after the execution of the Concession Agreement, as the case may be, any person in respect of any matter relating to the Project or the LOA or the Concession Agreement, who at any time has been or is a legal, financial or technical adviser of UTDB in relation to any matter concerning the Project;

b) "**Fraudulent practice**" means a misrepresentation or omission of facts or suppression of facts or disclosure of incomplete facts, in order to influence the Bidding Process;

(c) "**Coercive practice**" means impairing or harming or threatening to impair or harm, directly or indirectly, any person or property to influence any person's participation or action in the Bidding Process;

(d) "**Undesirable practice**" means (i) establishing contact with any person connected with or employed or engaged by UTDB with the objective of canvassing, lobbying or in any manner influencing or attempting to influence the Bidding Process; or (ii) having a Conflict of Interest; and

(e) "**Restrictive practice**" means forming a cartel or arriving at any understanding or arrangement among Bidders with the objective of restricting or manipulating a full and fair competition in the Bidding Process.

5.3

Signature of the Authorised Person ***On the Letterhead of the Bidder

Format for Covering Letter cum Project Undertaking

То

The Chief Executive Officer Uttarakhand Tourism Development Board, Pt.Deendayal Upadhaya Paryatan Bhawan, Near ONGC Helipad,Garhi Cantt. Dehradun-248 003

Dear Sir,

Re: Operation, Maintenance and Management of North Corbett Resort at Kotdwar, Uttarakhand

We have read and understood the Request for Proposal (RFP) Document in respect of the above Project downloaded from <u>www.uktenders.gov.in</u> and we hereby submit our Proposal.

We confirm that our Proposal is valid for a period of 120 (One Hundred Twenty) days from last date for submission of Proposals.

We hereby agree and undertake as under:

Notwithstanding any qualifications or conditions, whether implied or otherwise, contained in our Proposal we hereby represent and confirm that our Proposal is unqualified and unconditional in all respects and we agree to the terms of the Draft Concession Agreement, a draft of which also forms a part of the RFP Document provided to us.

Dated thisDay of, 2021.

Name of the Bidder

Signature of the Authorised Person

Name of the Authorised Person

Note:

• On the Letterhead of the Bidder.

Format for Anti-Collusion Certificate Anti-Collusion Certificate***

We hereby certify and confirm that in the preparation and submission of this Proposal, we have not acted in concert or in collusion with any other Bidder or other person/s and also not done any act, deed or thing which is or could be regarded as anti-competitive.

We further confirm that we have not offered nor will offer any illegal gratification in cash or kind to any person or agency in connection with the instant Proposal.

Name of the Bidder

Signature of the Authorised Person

Name of the Authorised Person

*** On the bidder's letter head.

(Performa of Performance Bank Guarantee)

THIS	DEED	OF	GUARANTEE	executed	on	this	the	day of
		at		by _				(Name of the
Bank) l	naving its	Head/l	Registered office at				h	ereinafter referred to as
"the Gu	uarantor"	which	expression shall ur	nless it be re	epugn	ant to	the su	bject or context thereof

In favour of

include successors and assigns;

Uttarakhand Tourism Development Board, Pt. Deendayal Upadhaya Paryatan Bhawan, Near ONGC Helipad, Garhi Cantt Dehradun- 248 003 hereinafter referred to as "UTDB", which expression shall, unless repugnant to the context or meaning thereof include its administrators, successors or assigns.

WHEREAS

By the Concession Agreement entered into between UTDB and______, a company incorporated under the provisions of the Companies Act, 2013, having its registered office/ permanent address at______ ("the Concessionaire"), the Concessionaire has been granted the Concession for **Operation**, **Maintenance and Management of North Corbett Resort at Kotdwar, Uttarakhand**, for a period of 30 years (hereinafter referred to as "the Project").

A. In terms of the Concession Agreement, the Concessionaire is required to furnish to UTDB, an unconditional and irrevocable bank guarantee for an amount of Rs.... (in words) as security for due and punctual performance/discharge of its obligations under the Concession Agreement, relating to Project by the Concessionaire.

At the request of the Concessionaire, the Guarantor has agreed to provide bank guarantee, being these presents guaranteeing the due and punctual performance/discharge by the Concessionaire of its obligations relating to the Project.

NOW THEREFORE THIS DEED WITNESSETH AS FOLLOWS:

Capitalised terms used herein but not defined shall have the meaning assigned to them respectively in the Concession Agreement.

1. The Guarantor hereby irrevocably guarantees the due and punctual performance by M/s._____ (hereinafter called "the Concessionaire") of all its obligations relating to the Project and in connection with achieving the project objectives by the Concessionaire in accordance with the Concession Agreement.

2. The Guarantor shall, without demur, pay to UTDB sums not exceeding in aggregate Rs. ... (in words) within seven (7) calendar days of receipt of a written demand thereof from UTDB stating that the Concessionaire has failed to meet its obligations under the Concession Agreement. The Guarantor shall not go into the veracity of any breach or failure on the part of the Concessionaire or validity of demand so made by UTDB and shall pay the amount specified in the demand, notwithstanding any direction to the contrary given or any dispute whatsoever raised by the Concessionaire or any other Person. The Guarantor's obligations hereunder shall subsist until all such demands are duly met and discharged in accordance with the provisions hereof.

In order to give effect to this Guarantee, UTDB shall be entitled to treat the Guarantor as the principal debtor. The obligations of the Guarantor shall not be affected by any variations in the terms and conditions of the Concession Agreement or other documents or by the extension of time for performance granted to the Concessionaire or postponement/non exercise/ delayed exercise of any of its rights by UTDB or any indulgence shown by UTDB to the Concessionaire and the Guarantor shall not be relieved from its obligations under this Guarantee on account of any such variation, extension, postponement, non-exercise, delayed exercise of any of its rights by UTDB, provided nothing contained wherein shall enlarge the Guarantor's obligation hereunder.

This Guarantee shall be irrevocable and shall remain in full force and effect until _____ unless discharged/ released earlier by UTDB in accordance with the provisions of the Concession Agreement. The Guarantor's liability in aggregate be limited to a sum of Rs. ... (in words).

3. This Guarantee shall not be affected by any change in the constitution or winding up of the Concessionaire/the Guarantor or any absorption, merger or amalgamation of the Concessionaire/the Guarantor with any other Person. The Guarantor has power to issue this guarantee and discharge the obligations contemplated herein, and the undersigned is duly authorised to execute this Guarantee pursuant to the power granted under _____.

IN WITNESS WHEREOF THE GUARANTOR HAS SET ITS HANDS HEREUNTO ON THE DAY, MONTH AND YEAR FIRST HEREINABOVE WRITTEN.

SIGNED AND DELIVERED by

Bank

by the hand of Shri ______and authorised official.

APPENDIX 4

Format for Technical Capability

(To be provided on the Letterhead of the Bidder and to be signed by the Bidder)

Previous Experience of the bidder:

Sl. No.	Financial Year	Location	Name of the Unit	Number of Rooms	Duration of Operation (in Months)

This is to certify that we have operated hotels with (cumulative) rooms for months during last 5 years immediately preceding bid due date.

.....

Name of the Bidder

Signature of the Authorised Person

Name of the Authorised Person

APPENDIX 5

Format for Financial Proposal

As per BoQ given in the Financial Folder.

APPENDIX 6 Format for Financial Capability

A certificate duly certified by chartered accountant along with his / her firm stamp and registration number as per following format will only be considered

Bidder	Net Worth (As on 31/03/2020)
Sole Bidder	
Lead Member of Consortium	
Other Member 1 (in case of Consortium)	
Other Member 2 (in case of Consortium)	
Total	

[{]Note:

- 1. The bidders shall submit detailed balance sheet with schedule of equity holding as a proof to establish non participation of associate companies in the bid process.
- 2. The Bidder shall provide the audited annual financial statements (Balance Sheet and Profit & Loss Account along with all schedules as required for this RFP Document.)
- 3. The Bidder should provide the Financial Capability based on its own financial statements. Financial capability of the Bidder's parent company or its subsidiary or any associate company (who are not members of the Consortium) will not be considered for computation of the Financial Capability of the Bidder.
- 4. Net Worth means:
 - For Company = Paid-up Equity Capital fund + Free Reserves
 - For Partnership Firm = Aggregate of partners' capital account + Reserves Aggregate of drawings by partners Aggregate of advances to partners (Revaluation reserves + Miscellaneous expenditure not written off)
 - For Proprietor firm/ Proprietorship = Capital account/ computation in ITR}

Power of Attorney for signing of Application

Know all men by these presents, We..... (name of the firm and address of the registered office) do hereby irrevocably constitute, nominate, appoint and authorise Mr/Ms (name), son/daughter/wife of and presently residing at, who is presently employed with us/ the Lead Member of our Consortium and holding the position of, as our true and lawful attorney (hereinafter referred to as the "Attorney") to do in our name and on our behalf, all such acts, deeds and things as are necessary or required in connection with or incidental to submission of our application for pre-qualification and submission of our bid for Operation, Maintenance and Management of North Corbett Resort at Kotdwar, Uttarakhand; Project proposed by Uttarakhand Tourism Development Board, (the "UTDB") including but not limited to signing and submission of all applications, bids and other documents and writings, participate in Pre-Applications and other conferences and providing information/ responses to UTDB, representing us in all matters before UTDB, signing and execution of all contracts including the Concession Agreement and undertakings consequent to acceptance of our bid, and generally dealing with UTDB in all matters in connection with or relating to or arising out of our bid for the said Project and/ or upon award thereof to us and/or till the entering into of the Concession Agreement with UTDB.

AND we hereby agree to ratify and confirm and do hereby ratify and confirm all acts, deeds and things done or caused to be done by our said Attorney pursuant to and in exercise of the powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Attorney in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us.

For (Signature, name, designation and address)

(Notarised)

Witnesses:

1. 2. Accepted

(Signature)

(Name, Title and Address of the Attorney)

Notes:

 \Box The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, laid down by the applicable law and the charter documents of the executant(s) and when it is so required, the same should be under common seal affixed in accordance with the required procedure.

□ Wherever required, the Applicant should submit for verification the extract of the charter documents and documents such as a board or shareholders' resolution/ power of attorney in favour of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Applicant.

□ For a Power of Attorney executed and issued overseas, the document will also have to be legalised by the Indian Embassy and notarised in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Applicants from countries that have signed the Hague Legislation Convention 1961 are not required to be legalised by the Indian Embassy if it carries a conforming Apostle certificate.

APPENDIX 8

Power of Attorney for Lead Member of Consortium

Whereas the Uttarakhand Tourism Development Board, Government of Uttarakhand has invited applications from eligible bidders for **Operation**, **Maintenance and Management of North Corbett Resort at Kotdwar**, **Uttarakhand** (the "Project")

Whereas,, and...., (collectively the "Consortium") being Members of the Consortium are interested in bidding for the Project in accordance with the terms and conditions of the Request for Proposal (RFP) and other connected documents in respect of the Project,

AND

Whereas, it is necessary for the Members of the Consortium to designate one of them as the Lead Member with all necessary power and authority to do for and on behalf of the Consortium, all acts, deeds and things as may be necessary in connection with the Consortium's bid for the Project and its execution.

NOW, THEREFORE, KNOW ALL MEN BY THESE PRESENTS

We hereby irrevocably authorise the Attorney (with power to sub-delegate) to conduct all business for and on behalf of the Consortium and any one of us during the bidding process and, in the event the Consortium is awarded the concession/contract, during the execution of the Project and in this regard, to do on our behalf and on behalf of the Consortium, all or any of such acts, deeds or things as are necessary or required or incidental to the pre-qualification of the Consortium and submission of its bid for the Project, including but not limited to signing and submission of all applications, bids and other documents and writings, participate in bidders and other conferences, respond to queries, submit information/ documents, sign and execute contracts and undertakings consequent to acceptance of the bid of the Consortium and generally to represent the Consortium in all its dealings with UTDB, and/ or any other Government Agency or any person, in all matters in connection with or relating to or arising out of the Consortium's bid for the Project and/ or upon award thereof till the Concession Agreement is entered into with UTDB.

AND hereby agree to ratify and confirm and do hereby ratify and confirm all acts, deeds and things done or caused to be done by our said Attorney pursuant to and in exercise of the powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Attorney in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us/ Consortium.

IN WITNESS WHEREOF WE THE PRINCIPALS ABOVE NAMED HAVE EXECUTED THIS POWER OF ATTORNEY ON THIS DAY OF, 2021

For

UTDB

(Signature)

(Name & Title) For (Signature)

.....

(Name & Title) For

(Signature)

(Name & Title)

(Notarised)

Witnesses: 1. 2. (Executants) (To be executed by all the Members of the Consortium)

Notes:

- 1. The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, lay down by the applicable law and the charter documents of the executants(s) and when it is so required, the same should be under common seal affixed in accordance with the required procedure.
- 2. Also, wherever required, the Bidder should submit for verification the extract of the charter documents and documents such as a board or shareholders' resolution/ power of attorney in favour of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Bidder.
- 1. For a Power of Attorney executed and issued overseas, the document will also have to be legalised by the Indian Embassy and notarised in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Bidders from countries that have signed The Hague Legislation Convention, 1961 are not required to be legalised by the Indian Embassy if it carries a conforming Apostle certificate.

APPENDIX 9

Joint Bidding Agreement

(To be executed on Stamp paper of appropriate value)

THIS JOINT BIDDING AGREEMENT is entered into on this the day of, 2021

AMONGST

1. {..... Limited, a company incorporated under the Companies Act, 2013} and having its registered office at (hereinafter referred to as the "First Part" which expression shall, unless repugnant to the context include its successors and permitted assigns)}

AND

2. {..... Limited, a company incorporated under the Companies Act, 2013} and having its registered office at (hereinafter referred to as the "Second Part" which expression shall, unless repugnant to the context include its successors and permitted assigns)}

AND

3. {..... Limited, a company incorporated under the Companies Act, 2013 and having its registered office at (hereinafter referred to as the "Third Part" which expression shall, unless repugnant to the context include its successors and permitted assigns)}

WHEREAS

The Parties are interested in jointly bidding for the Project as members of a Consortium and in accordance with the terms and conditions of the RFP document and other bid documents in respect of the Project, and It is a necessary condition under the RFP document that the members of the Consortium shall enter into a Joint Bidding Agreement and furnish a copy thereof with the Application.

NOW IT IS HEREBY AGREED as follows:

1. Definitions and Interpretations

a) In this Agreement, the capitalised terms shall, unless the context otherwise requires, have the meaning ascribed thereto under the RFP.

2. Consortium

- a) The Parties do hereby irrevocably constitute a consortium (the "Consortium") for the purposes of jointly participating in the Bidding Process for the Project.
- b) The Parties hereby undertake to participate in the Bidding Process only through this Consortium and not individually and/ or through any other consortium constituted for this Project, either

directly or indirectly or through any of their Associates.

3. Covenants

a) The Parties hereby undertake that in the event the Consortium is declared the Successful Bidder and awarded the Project, it shall incorporate a special purpose vehicle (the "SPV") under the Companies Act, 2013 for entering into a Concession Agreement with UTDB and for performing all its obligations as the Concessionaire in terms of the Concession Agreement for the Project.

4. Role of the Parties

The Parties hereby undertake to perform the roles and responsibilities as described below:

- a) Party of the First Part shall be the Lead member of the Consortium and shall have the power of attorney from all Parties for conducting all business for and on behalf of the Consortium during the Bidding Process and until the Appointed Date under the Concession Agreement when all the obligations of the SPV shall become effective;
- b) Party of the Second Part shall be the (Technical) Member of the Consortium;
- c) Party of the Third Part shall be the (Financial) Member of the Consortium;
- d) Party of the Fourth Part shall be the (Operation and Management) Member/ Other Member of the Consortium.

5. Joint and Several Liabilities

a) The Parties do hereby undertake to be jointly and severally responsible for all obligations and liabilities relating to the Project and in accordance with the terms of the RFP and the Concession Agreement, till such time as the Commissioning for the Project is achieved under and in accordance with the Concession Agreement.

6. Shareholding in the SPV

- a) The Parties agree that the proportion of shareholding among the Parties in the SPV shall be as follows:
 - First Party:
 - Second Party:

b) The Parties undertake that a minimum of 51% (Fifty One per cent) of the subscribed and paid up equity share capital of the SPV shall, at all times till the concession period, be held by the First Party whose experience and net worth has been reckoned for the purposes of qualification for the Project in terms of the RFP

- c) The Parties undertake that they shall collectively hold 100% (One Hundred per cent) of the subscribed and paid up equity share capital of the SPV at all times until end of the Concession period.
- d) The Parties undertake that they shall comply with all equity lock-in requirements set forth in the Concession Agreement.

7. Representation of the Parties

Each Party represents to the other Parties as of the date of this Agreement that:

- (a) Such Party is duly organised, validly existing and in good standing under the laws of its incorporation and has all requisite power and authority to enter into this Agreement;
- (b) The execution, delivery and performance by such Party of this Agreement has been authorised by all necessary and appropriate corporate or governmental action and a copy of the extract of the charter documents and board resolution/ power of attorney in favour of the person executing this Agreement for the delegation of power and authority to execute this Agreement on behalf of the Consortium Member is annexed to this Agreement, and will not, to the best of its knowledge:
 - i. require any consent or approval not already obtained;
 - ii. violate any Applicable Law presently in effect and having applicability to it;
 - iii. violate the memorandum and articles of association, by- laws or other applicable organisational documents thereof; iv. violate any clearance, permit, concession, grant, license or other governmental authorisation, approval, judgement, order or decree or any

mortgage agreement, indenture or any other instrument to which such Party is a party or by which such Party or any of its properties or assets are bound or that is otherwise applicable to such Party; or v. create or impose any liens, mortgages, pledges, claims, security interests, charges or Encumbrances or obligations to create a lien, charge, pledge, security interest, encumbrances or mortgage in or on the property of such Party, except for encumbrances that would not, individually or in the aggregate, have a material adverse effect on the financial condition or prospects or business of such Party so as to prevent such Party from fulfilling its obligations under this Agreement;

- (c) This Agreement is the legal and binding obligation of such Party, enforceable in accordance with its terms against it.
- (d) There is no litigation pending or, to the best of such Party's knowledge, threatened to which it or any of its Affiliates is a party that presently affects or which would have a material adverse effect on the financial condition or prospects or business of such Party in the fulfilment of its obligations under this Agreement.

8. Termination

- a) This Agreement shall be effective from the date hereof and shall continue in full force and effect until the Commercial operation of the Project is achieved under and in accordance with the Concession Agreement, in case the Project is awarded to the Consortium.
- b) However, in case the Consortium is either not pre-qualified for the Project or does not get selected for award of the Project, the Agreement will stand terminated in case the Bidder is not prequalified or upon return of the Bid Security by UTDB to the Bidder, as the case may be.

9. Miscellaneous

This Joint Bidding Agreement shall be governed by laws of India. The Parties acknowledge and accept that this Agreement shall not be amended by the Parties without the prior written consent of UTDB.

IN WITNESS WHEREOF THE PARTIES ABOVE NAMED HAVE EXECUTED AND DELIVERED THIS AGREEMENT AS OF THE DATE FIRST ABOVE WRITTEN.

SIGNED, SEALED AND DELIVERED For and on behalf of:

LEAD MEMBER by: (Signature) (Name) (Designation) (Address)

SIGNED, SEALED AND DELIVERED For and on behalf of: SECOND PART by: (Signature) (Name) (Designation) (Address)

SIGNED, SEALED AND DELIVERED For and on behalf of: THIRD PART by: (Signature) (Name) (Designation) (Address)

(Notarised)

In the presence of: 1) 2)

Notes:

- 1. The mode of the execution of the Joint Bidding Agreement should be in accordance with the procedure, if any, lay down by the Applicable Law and the charter documents of the executants(s) and when it is so required, the same should be under common seal affixed in accordance with the required procedure.
- 2. Each Joint Bidding Agreement should attach a copy of the extract of the charter documents and documents such as resolution / power of attorney in favour of the person executing this Agreement for the delegation of power and authority to execute this Agreement on behalf of the Consortium Member.
- 3. For a Joint Bidding Agreement executed and issued overseas, the document shall be legalised by the Indian Embassy and notarised in the jurisdiction here the Power of Attorney has been executed.

PART II - DRAFT CONCESSION AGREEMENT

BETWEEN

Uttarakhand Tourism Development Board. ["CONCESSIONING AUTHORITY"]

AND

(Name of the Concessionaire) ["CONCESSIONAIRE"]

<u>{This Contract will be suitably modified as per changes that may be required / relevant</u> <u>based on Proposal finally accepted and pre-contract negotiation</u>)

ARTICLE	CONTENTS	Page no.
1	Definitions & Intermetations	40
1	Definitions & Interpretations	40
2	The Concession	48
3	Obligations of Concessionaire	52
4	Obligations of Concessioning Authority	57
5	Deleted	58
6	Force Majeure	59
7	Events of Default and Termination	63
8	Dispute Resolution	69
9	Financing and Lender's Rights	71
10	Representations, Warranties and Disclaimer	72
11	Miscellaneous	74
SCHEDULE		
1	Project Site including details of Existing Infrastructure	77
2	Concession Payment Schedule	78
3	Scope of Construction & Minimum Construction Requirement	79
4	Operation & Maintenance Requirements	80
5	Performance Security	81
6	Vesting Certificate	83
7	Deleted	84
8	License Agreement	85
9	Draft Project Site Lease Deed	90

TABLE OF CONTENTS

This AGREEMENT is entered into on this the _____ day of _____ (month), _____ (year) at Dehradun,

BETWEEN

Uttarakhand Tourism Development Board with its registered office at **Pt. Deendayal Upadhyaya Paryatan Bhawan, (Near ONGC Helipad), Garhi Cantt., Dehradun – 248 003 represented by its Chief Executive Office** (hereinafter referred to as "**Concessioning Authority**", which expression shall, unless the context otherwise requires, includes its administrators, successors and assigns) of the **ONE PART**;

AND

_____Pvt. Ltd., a company incorporated under provisions of the Companies Act, 1956, having its registered office at _____, (or M/s _____ duly registered Partnership/Proprietorship Firm). (hereinafter referred to as the "Concessionaire", which expression shall, unless otherwise requires, mean and includes its successors/ successors in business and permitted assigns and substitutes) of the OTHER PART.

WHEREAS,

- A. Concessioning Authority has decided to select the Successful Bidder through a competitive bidding process to Install, Operate, Maintain and manage the facilities listed in Annexure "A" on Public Private Partnership (PPP) basis (the "**Project**") at Kotdwar, which shall be given by Concessioning Authority to the Concessionaire.
- B. Concessioning Authority is in the possession of the plots, which shall be given by Concessioning Authority to the Concessionaire (as per details of Project Site attached at Schedule 1).
- C. With an objective to seek private sector participation in the development of the aforesaid Project, Concessioning Authority had invited competitive Proposals/ Bids from interested parties to partner Concessioning Authority *inter alia*, to subsequently operate and maintain the tent colonies including marketing the same on the terms and conditions contained in the Request for Proposal (RFP) document and to determine, levy, demand, collect, retain and appropriate User Charges from the Users of the Project Facility;
- D. In response to the Request for Proposal document, as referred to in recital 'B' above, the Concessioning Authority received Proposals from various Bidders;
- E. Concessioning Authority evaluated the Financial Proposals submitted by all Bidders who met the Technical Qualification Criterion specified in the RFP document and accepted the Proposal/Bid submitted by the **"Successful Bidder"** consisting of M/s ______, ____& _____(collectively the **"Consortium**) with ______ as its Lead Member or M/s _______ (Single/ Individual Bidder) and a Notice of Award, bearing No. ______ Dated______ was issued to the Successful Bidder.
- F. Accordingly, as per the terms and conditions of the RFP Document, the Successful Bidder has / have incorporated the Concessionaire as a Special Purpose Company in India, under the Companies Act, 1956/2013 to implement the Project.
- H. In pursuance to the above, Concessioning Authority is executing this Agreement to grant the Concession to the Concessionaire for operation and maintenance of tent colonies, including, as applicable, design, engineer, finance, construct, market and manage the above mentioned Project and its allied facilities at Kotdwar, Uttarakhand, in accordance with terms, conditions and covenants hereinafter set forth.

NOW THEREFORE THIS AGREEMENT WITNESSETH AND IT IS HEREBY AGREED BY AND BETWEEN THE PARTIES HERETO AS FOLLOWS:

ARTICLE 1 DEFINITIONS & INTERPRETATIONS

1.1 Definitions

In this Agreement, except to the extent that the subject or context otherwise requires, the following expressions shall have the following meanings:

"Affected Party" shall have the meaning ascribed to it in Article 6.1.1.

"Affiliate" means, with respect to any Party and/or with respect to the Single/ Individual Bidder and/or with respect to any member of Consortium, any other Person directly or indirectly controlling, controlled by or under common control with such Party, Single/ Individual Bidder and/or member of Consortium. For the purposes of this definition, the term "control" (including with correlative meaning, the terms "controlled by" and "under common control with") as applied to any Party or Single/ Individual Bidder or a member of Consortium, means the possession, directly or indirectly, of the power to direct or cause the direction of the management of that Party or Single/ Individual Bidder or a member of Consortium whether through ownership of 50% (fifty percent) or more of the voting securities, by contract, or otherwise.

"Agreement" means this signed Concession Agreement (including the Schedules, Recitals of the Concession Agreement, the Notice of Award issued by Concessioning Authority, the written clarifications, addendums, amendments etc. to the RFP document issued subsequently to the Bidders and all other documents and papers attached as annexure).

"Annual Concession Fee" shall have the meaning ascribed to it in Article 2.9.1.

"Appendix" means any of the schedules, supplements or documents appended to this Agreement.

"Applicable Laws" means any statute, law, regulation, development control regulations, ordinance, notification, rule, judgment, order, decree, bye-law, clearance, directive, guideline, policy, requirement, or other governmental restriction or any similar form of decision of, or determination by, or any interpretation or administration having the force of law of any of the foregoing, by any Government Agency or instrumentality thereof having jurisdiction over the matter in question, as may be in force and effect during the subsistence of this Agreement.

"Applicable Permits" means all clearances, permits, authorisations, permissions, consents, exemptions, licenses, no-objection certificates and approvals required to be obtained or maintained under or pursuant to Applicable Laws, in connection with implementation of the Project during the subsistence of this Agreement.

"Appointed Date" means the date of this Agreement.

"Arbitration Act" means the Arbitration and Conciliation Act, 1996 and shall include modifications to or any re-enactment thereof as in force from time to time.

"Certificate of Compliance" shall have the meaning ascribed to it in Article 2.4.1(ii).

"Clearance" means, as on the date of execution of this Agreement, any consents, licenses, approvals, permits, exemptions, registrations, filings or other authorizations of whatever nature, which is necessary for effective implementation of the Project.

"Concession Period" or "Term" of the Concession Agreement shall have the meaning ascribed to it in Article 2.7.

"Concessionaire Event of Default" shall have the meaning ascribed to it in Article 7.1 (a).

"Concessionaire's Proposal to Rectify" shall have the meaning ascribed to it in Article 7.2 (a) (ii).

"Concessioning Authority Event of Default" shall have the meaning ascribed to it in Article 7.1 (b).

"**Concessioning Authority Proposal to Rectify**" shall have the meaning ascribed to it in Article 7.2 (b) (ii).

"Consortium" means the consortium consisting of (i) Member 1, (ii) Member 2, and (iii) Member 3 formed, to implement the Project

"Competent Authority" means the concerned department, Government Instrumentality or authority of the Concessioning Authority duly empowered and authorized to grant necessary sanctions and approvals.

"Compliance Date" means the date falling after 60 (Sixty) days after the Appointed Date or such other date as agreed to between Concessioning Authority and the Concessionaire by which time the Parties fulfill their Conditions Precedent (under Article 2.4), by the issuance of "Certificate of Compliance" to each other by the respective Parties / their authorized representatives.

"Conditions Precedent" shall have the meaning ascribed to it in Article 2.4.

"Contractor(s)" means a reputed Person with whom the Concessionaire has entered into or may enter into contracts / agreements for the purpose of development along with the operation and maintenance of the Project.

"Cure Period" shall have the meaning ascribed to it in Article 7.2 (a) (iii).

"Debt Due" means the aggregate of the following sums expressed in Indian Rupees outstanding and payable to the Senior Lender under the Financing Documents: -

- a) The principal amount of the debt provided by the Senior Lender under the Financing Documents for financing the Project, which is outstanding as on the Termination Date, excluding any part of the principal that had fallen due for repayment one year prior to the Termination Date unless such repayment had been rescheduled with the prior consent of the Concessioning Authority; and
- b) All accrued interest, financing fee and charges payable on or in respect of the debt referred to in the Para (a) above up to the date preceding the Termination Date but excluding any interest or charges that had fallen due one year prior to the Termination Date, and penal interest or charges payable under the Financing Documents to any Senior Lender.

"Dispute Resolution" shall have the meaning ascribed to it in Article 8.

"Easementary Rights" means all easements, reservations, right-of-way, utilities and other similar rights as to the use of the real property, which are necessary or appropriate for the conduct of business of the Concessionaire related to the Project.

"Encumbrance" means any encumbrance such as mortgage, charge, pledge, lien, hypothecation, security, interest, assignment, privilege or priority of any kind having the effect of security or other such obligations and shall include without limitation any designation of loss payees or beneficiaries or any similar arrangement under any insurance policy pertaining to the Project, physical encumbrances and encroachments on the Project Site.

"Expert" shall have the meaning ascribed to it in Article 5 and Schedule 7.

"Financial Assistance" means the aggregate amount provided to the Concessionaire by way of loan, lease finance, advances, guarantees, refinancing or otherwise and other debt instruments by the Senior Lenders for the financing of the Total Project Cost (including any amendments or modifications made thereto) and for all facilities and services relating to the Project.

"Financial Closure" means the date on which the Financing Documents with respect to the Financing Package for the Project have been executed and become effective and the Concessionaire has fulfilled all the conditions needed for draw down of Financing and the Concessionaire has immediate access to such Financing and which shall in any case not be later than 180 days from the Appointed Date as applicable hereof..

"Financial Year" means the year commencing from 1st April of any calendar year to the 31st March of the next calendar year except in the first and the last calendar year of the subsistence of this Agreement. In the first year of subsistence of this Agreement, it means the period from the Compliance Date to immediately following 31st March. In the last year of subsistence of this agreement, it means the period from 1st April to the Transfer Date.

"Financing Documents" means the documents / agreements executed by the Concessionaire in respect of Financial Assistance to be provided by the Senior Lenders to the Concessionaire by way of loans, guarantees, subscription to non-convertible debentures and other debt instruments including loan agreements, guarantees, notes, debenture bonds and other debt instruments, security agreements and other documents relating to the financing (including refinancing) of the Total Project Cost.

"Force Majeure" or "Force Majeure Event" shall have the meaning ascribed to it in Article 6.

"Good Industry Practice" means the exercise of that degree of skill, diligence, prudence and foresight in compliance with the undertakings and obligations under this Agreement which would reasonably and ordinarily be expected from a skilled and experienced Person engaged in the development, implementation, operation and maintenance or supervision or monitoring thereof or any of them of a Project of the type and size similar to that of the Project.

"GoI" shall mean the Government of India.

"Legal Entity" shall mean anybody corporate, organization, society, trust or institution established as per the applicable laws of India, or in case of a Foreign Entity, recognized or found eligible by Government of India to enter into any commercial trade or transaction, after obtaining necessary approvals and sanctions from the Government of India in this regard, as per the applicable laws of India including national security and public interest perspective.

"Material Adverse Effect" means consequences of events outside the control of the Affected Party which (a) render any right vested in a Party by the terms of this Concession ineffective, or (b) significantly impairs or frustrates the ability of any Party to observe and perform in a timely manner its obligations under this Concession Agreement, or (c) frustrates a material provision of this Concession Agreement or any of the Project Agreements.

"Material Breach" means a breach by either Party of any of its obligations under this Agreement which has or is likely to have a Material Adverse Effect on the Project and which such Party shall have failed to cure.

"Movable Assets" shall mean and include all such machines, equipment and furniture of the Project Facility, which are tangible in nature and that is movable (and not immovable like the civil structure etc), and which was owned by the Concessionaire and used as part of overall development (especially during the Operation & Maintenance) of the Project Facility.

"CONCESSIONAIRE" means Successful bidder, who has entered into contract for this Project.

"Notice to Commence" shall have the meaning ascribed to it in Article 2.4.1 (iii).

"Operation and Maintenance Phase" is the period commencing after the execution of Concession Agreement Completion Date (as per Project Milestone in Article 2.8) and terminating at the Transfer Date.

"O&M Manual" shall have the meaning ascribed to it in Article 3.5 (b)

"O&M Expenses" means expenses incurred towards Operations and Maintenance of the Project Facility/ies

"Parties" means the parties to this Agreement and "Party" shall mean either of them, as the context may admit or requires.

"Performance Security" means the Performance Security by way of Bank Guarantee from a scheduled bank acceptable to Concessioning Authority, as per Schedule 5, and as also ascribed to in Article 3.2.

"**Person**" unless specifically provided otherwise, shall mean any individual, company, corporation, partnership, joint venture, trust or Government Agency or any other legal entity as the context may admit.

"Preliminary Notice to Remedy" shall have the meaning ascribed to it in Article 7.2 (a) (ii) and Article 7.2 (b) (ii).

"Project" means subject to the provisions of this Concession Agreement, the operation and maintenance of existing infrastructure and proposed development including design, engineer, finance, construct, market and manage, at the Project Site, of (i) Facilities identified; (ii) levy, collection, retention and appropriation of User Charges from Users of the Project Facility; (iii) insurance of the Project Facility, for the purposes of providing the services on a continuous basis and (iv) transfer of the Project Facility, as envisaged, to the Concessioning Authority at the end of the Concession Period or on prior termination of the Concession Agreement on Concessionaire/ Concessioning Authority Event of Default or otherwise. The Project shall also include all equipment, foundations, buildings, structures, super-structures, constructions, additions, alterations or improvements, etc. thereof, landscape structures, pavements, walkways, drainage facilities, sign boards, electrical, mechanical, civil, sanitation, HVAC, furnishings and other works, equipment installed therein to be used for the purposes of the Project and including but not limited to the furniture, fixtures and equipment, other related facilities installed or affixed and all additions, modifications, alterations and extensions thereto as may be effected by the Concessionaire from time to time, as per the approved and sanctioned plans on the Project Site.

"Project Agreements" means collectively, this Concession Agreement, the Financing Documents, Construction Agreements and Operation & Maintenance Agreements, in each case as amended, supplemented or otherwise modified from time to time and any other agreements or contract that may be entered into by the Concessionaire with any person in connection with matters relating to, arising out of or incidental to the Project.

"Project Assets" means all tangible and intangible assets relating to the Project / Project Facility.

"Project Construction Completion Date" – means 2 (two) months from the date of issue of compliance certificate.

"Project Facility" means facilities and amenities to be provided in the hotels as per the specifications given under Schedule 3.

"Project Milestone" means the milestone as set out in Article 2.8.

"Project Milestone Completion Certificate" means the certificate issued to the Concessionaire by the Concessioning Authority as described in Article 2.8 and Article 2.8.1.

"Project Site" means the land area at Kotdwar, which is covered and secured by boundary wall with existing infrastructure. However, the details of Project Site has been laid down in Schedule-1 and Schedule 1 (B), wherein the Project/ Project Facility is to be developed, constructed, implemented, established, operated, maintained and managed by the Concessionaire in accordance with the provisions of this Agreement.

"Regulatory Approvals" shall mean Governmental or regulatory approvals required by the Parties for the establishment of the Project in Kotdwar and as per the applicable laws of India.

"**Reference Exchange Rate**" means, in respect of any one currency that is to be converted into another currency in accordance with the provisions of this Agreement, the exchange rate as of 12.00 (twelve) noon on the relevant date quoted in Delhi by the State Bank of India;

"Rupee(s)" or "Rs." or "Re." shall mean Indian rupee(s).

"SBI PLR" means the prime lending rate per annum for loans with 1 (one) year maturity as fixed from time to time by the State Bank of India or any other arrangement that substitutes such prime lending rate as mutually agreed between the Parties.

"Senior Lenders" means financial institutions, banks and multilateral lending agencies including their successors and assigns, who have agreed to guarantee or provide finance to the Concessionaire under any of the Financing Documents for meeting all or any part of the Total Project Cost and who hold *parri pasu* charge on the assets, rights, title and interests of the Concessionaire

"Structural Engineer" shall have the meaning ascribed to it in Article 3.4 (g).

"Structural Safety Certificate" shall have the meaning ascribed to it in Article 3.4 (e).

"Subordinated Debt" means the aggregate of the following sums expressed in Indian Rupees or in the currency of debt, as the case may be, outstanding as on the Transfer Date:

- (a) the principal amount of debt provided by lenders for the Concessionaire for meeting the Total Project Cost and subordinated to the Financial Assistance provided by the Senior Lenders; and
- (b) all accrued interest on the debt referred to in Sub-clause (a) above but restricted to the lesser of actual interest rate and a rate equal to 5% (five per cent) above the SBI PLR in case of loans expressed in Indian Rupees and lesser of the actual interest rate and six-month LIBOR (London Inter-Bank Offer Rate) plus 2% (two per cent) in case of loans expressed in foreign currency, but does not include any interest that had fallen due one year prior to the Transfer Date; provided that if all or any part of the Subordinated Debt is convertible into Equity at the option of the lenders and/or the Concessionaire, it shall for the purposes of this Agreement be deemed to be Subordinated Debt even after such conversion and the principal thereof shall be dealt with as if such conversion had not been undertaken;

"Successful Bidder" means the Single/ Individual Bidder or the Bidder Consortium consisting of M/s _____, ____ & _____ (with _____ as its Lead Member) that is finally awarded the Project and invited to enter into this Concession Agreement.

"Termination" means early termination of this Agreement, pursuant to Termination Notice or non-fulfilment of Conditions Precedent or otherwise in accordance with the provisions of this Agreement but shall not, unless the context otherwise requires, include expiry of this Agreement due to efflux of time in the normal course.

"Termination Date" means the date specified in the Termination Notice as the date on which Termination occurs.

"Termination Notice" means the notice of Termination of this Agreement by either Party to the other Party, in accordance with the applicable provisions of this Agreement.

"Termination Payment" means the amount payable by the Concessioning Authority to the Concessionaire upon Termination in accordance with the provisions of this Agreement; It is agreed that within a period of 60 (sixty) days from Project Construction Completion Date, the Concessionaire shall notify to the Concessioning Authority, the Total Project Cost as on Project Construction Completion Date and its disaggregation between Debt Due, Subordinated Debt and Equity, and only the amounts so conveyed shall form the basis of computing Termination Payment, and it is further agreed that in the event such disaggregation is not notified to the Concessioning Authority, Equity shall be deemed to be the product arrived at by subtracting Debt Due from Total Project Cost.

"Third Party" or "Third Parties" means any Person, real or legal or Entity other than the Parties of this Agreement.

"Total Project Cost" means the lowest of the following:

- (a) Total Project Cost as set forth in the Financing Documents.
- (b) Actual capital cost of the Project upon completion of the Project as certified by statutory auditor(s).

"Transfer Assets" shall mean both the Movable Assets and immovable assets, fully operational and functional, of the Project which are necessary or required for the performance of services and such other assets as Concessionaire procures in accordance with the provisions of this Agreement and shall specifically include all land, property and structures thereupon acquired during the term, all equipment and services, furnishings, etc. in relation to the operation of the Project, as existing on the date of Termination.

"Transfer Date" means the date immediately following the date of the expiry of the Term under this Concession Agreement or any earlier termination thereof in accordance with the provisions of this Agreement.

""Users" means all the persons who use all or any one or more of the facilities, amenities, utilities etc. established or provided in the Project under suitable arrangement entered into by such persons with the Concessionaire / Concessionaire's sub-lessees on terms and conditions as stipulated from time to time by the Concessionaire's sub-lessees in this regard.

"User Charges" means all charges, costs, fees, tariff, rentals and other amounts by whatever name called, payable to Concessionaire by the users or any other Person, etc. pursuant to this Agreement, for the purpose of rendering of services by the Concessionaire and utilisation of the Project Facility of the Project by such users, or other Persons, at any time and from time to time in relation to the Project.

"Vacant Possession" means delivery of the possession of the Project Site including the existing structures and development work free from all Encumbrances, restrictions or impediments and the grant of all Easements and all other rights appurtenant or in relation thereto.

"Vesting Certificate" shall have the meaning ascribed to it in Article 7.6 (viii).

1.2 Interpretation

In this Agreement, unless the context otherwise requires:

- a. any reference to a statutory provision shall include such provision as is from time to time modified or re-enacted or consolidated so far as such modification or re-enactment or consolidation applies or is capable of applying to any transactions entered into hereunder;
- b. references to Applicable Law shall include the laws, acts, ordinances, rules, regulations, notifications, guidelines or byelaws which have the force of law in Uttarakhand forming part of the Union of India;
- c. the words importing singular shall include plural and vice versa, and words denoting natural persons shall include partnerships, firms, companies, corporations, joint ventures, trusts, associations or partnerships, organisations or other entities (whether or not having a separate legal entity);
- d. terms and words beginning with capital letter and defined in this Agreement shall have the meaning ascribed thereto herein, and terms and words defined in the Schedule and used therein shall have the meaning ascribed thereto in the Schedule;
- e. the headings are for convenience of reference only and shall not be used in, and shall not affect, the construction or interpretation of this Agreement;
- f. the words "include" and "including" are to be construed without limitation;
- g. references to "construction" include investigation, design, engineering, procurement, delivery, transportation, installation, processing, fabrication, testing, commissioning and other activities incidental to the construction;
- h. any reference to any period of time shall mean a reference to that according to Indian Standard Time;
- i. any reference to day shall mean a reference to a calendar day;
- j. any reference to month shall mean a reference to a calendar month as per the Gregorian Calendar;
- k. reference to an individual shall include his legal representatives, successors, legal heirs, executor and administrator;
- 1. the Schedules to this Agreement form an integral part of this Agreement and will be in full force and effect as though they were expressly set out in the body of this Agreement;
- m. any reference at any time to any agreement, deed, instrument, license or document of any description shall be construed as reference to that agreement, deed, instrument, license or other document as amended, varied, supplemented, modified or suspended at the time of such reference;

- n. references to recitals, Articles, sub-articles or Schedules in this Agreement shall, except where the context otherwise requires, be deemed to be references to recitals, Articles, sub-articles and Schedules of or to this Agreement;
- o. any agreement, consent, approval, authorisation, notice, communication, information or report required under or pursuant to this Agreement from or by any Party shall be valid and effectual only if it is in writing under the hands of duly authorised representative of such Party, in this behalf and not otherwise; and
- p. unless otherwise stated, any reference to any period commencing "from" a specified day or date and "till" or "until" a specified day or date shall include both such days or dates.

1.3 Measurements and Arithmetic Conventions

All measurements and calculations shall be in metric system and calculations done to 2 decimal places, with the third digit of 5 or above being rounded up and below 5 being rounded down.

1.4 Ambiguities and Discrepancies

In case of ambiguities or discrepancies within this Agreement, the following shall apply:

- a. between two Articles of this Agreement, the provisions of specific Articles relevant to the issue under consideration shall prevail over those in other Articles;
- b. between the Articles of this Agreement and the Schedules, the Articles shall prevail, between Schedules and Appendices, the Schedules shall prevail;
- c. between the written description on the drawings and the specifications and standards, the latter shall prevail;
- d. between the dimension scaled from the drawing and its specific written dimension, the latter shall prevail; and
- e. Between any value written in numerals and that in words, the latter shall prevail.

1.5 Priority of Documents

The documents forming this Concession Agreement are to be taken as mutually explanatory of one another. If there is an ambiguity or discrepancy in the documents, the Concessioning Authority shall issue any necessary clarification or instruction to the Concessionaire, and the priority of the documents shall be as follows:

- i) This signed Concession Agreement
- ii) Instruction to Bidder

ARTICLE 2 THE CONCESSION

2.1 Grant of rights to Concessionaire

- a. Subject to and in accordance with the terms and conditions set forth in this Agreement, the Applicable Laws and Applicable Permits, the Concessioning Authority hereby authorises the Concessionaire to invest, finance, design, engineer, construct, equip, commission, operate, market, maintain and manage the Project during the Term of this Agreement.
- b. For the purpose of implementing the Project, Concessioning Authority *inter alia* grants to the Concessionaire all the rights, powers, benefits, privileges, authorisations and entitlements, to utilise the Project Site for the purposes of the Project, and to further undertake the development, construction and improvement therein or thereon, as may be necessary or appropriate to manage the Project and the right to demand and collect User Charges from the Users of the Project / Project Site in accordance with the provisions of this Agreement.
- c. Subject to the terms of this Agreement and other relevant provisions under Applicable Laws, the Concessionaire shall have the right to enter into agreements with such Persons, as it may deem necessary and appropriate, for performing its obligations under this Agreement.
- d. The Concessionaire may create an Encumbrance, with respect to its rights, in favour of Senior Lenders, only during the Lease Period with prior written approval of the Concessioning Authority.
- e. During the Concession Period, the Concessionaire shall have exclusive authority to operation, maintenance, develop, implement the Project and in accordance with the provisions of this Agreement.

2.2 Acceptance by Concessionaire

In consideration of the rights, privileges and benefits conferred upon by Concessioning Authority and other good and valuable consideration expressed herein, the Concessionaire hereby accepts and agrees and undertakes to perform / discharge all of its obligations in accordance with the provisions hereof.

2.3 Transfer of Project Site

Within 7 (Seven) days from the Appointed Date, Concessioning Authority shall hand over the Project site. The infrastructure already developed by Concessioning Authority as per detail layout plan and peaceful physical possession of the Project Site and existing structures to Concessionaire on 'as-is-where-is' basis, free from all Encumbrances.

2.4 Conditions Precedent

a. Save and except as may otherwise be expressly provided herein, the respective rights and obligations of the Parties under this Agreement shall be subject to the satisfaction in full of the conditions precedent ("Conditions Precedent") as set out in Article 2.4 (b) and Article 2.4 (c) on or before the expiry of a period of 60 (Sixty) days from the Appointed Date.

b. Conditions Precedent for the Concessioning Authority

The Concessioning Authority shall have:

- i. Delivered developed & secured by permanent boundary wall area of the Project Site at the time of execution of the Project Site Lease Deed (in the form attached hereto in Schedule 9 and co-terminus with this Concession Agreement);
- ii. Give comments/ observations (if any) to the concept plan submitted by the Concessionaire under Article 2.4(c) (i), within a time period of 30 (Thirty) days from the date of its receipt. Failure to do so will be considered deemed approval of the concept plan by the Concessioning Authority.
- iii. Give approval to the amended concept plan within 30 (Thirty) days from the date of its receipt.

c. Conditions Precedent for the Concessionaire

The Concessionaire shall have:-

- i. Prepared a concept plan for the Project and preliminary project cost estimates, including the area statement, block drawings, details of services to be so provided and as incorporated in the concept plan as per the applicable Local Byelaws, Norms, etc and submit to the Concessioning Authority for its approval, whereby meeting all the requirements mentioned herein, apart from all other requirements of this Concession Agreement in the concept plan within a period of 30 days from the Appointed Date.
- ii. Incorporated the necessary suggestions / amendments proposed by the Concessioning Authority in its concept plan, within a period of 10 (ten) days from the date of receipt of such suggestions from the Concessioning Authority and submit again for approval.
- iii. Provided the Concessioning Authority copies (certified as true by the Director of the Concessionaire) of all resolutions adopted by the Board of Directors of the Concessionaire authorizing the execution, delivery and performance of this Concession Agreement by the Concessionaire.
- iv. Delivered to the Concessioning Authority, a legal opinion from the legal counsel of the Concessionaire with respect to the authority of the Concessionaire to enter into this Agreement and the enforceability of the provisions thereof.
- v. Executed and procured execution of the Substitution Agreement.
- vi. Executed the Financing Agreements and delivered to the Concessioning Authority, 3 (three) true copies thereof, duly attested by a Director of the Concessionaire.
- vii. Executed an agreement to operate the facility for entire concession period with any International / National Hotel operator chain which is running at least one property of 3 star category or above within India or Abroad

2.4.1 Obligations to Satisfy Conditions Precedent

- i. Each Party hereto shall use all reasonable endeavours at its cost and expense to procure the satisfaction in full of its respective Conditions Precedent set out above within 60 (Sixty) days of Appointed Date.
- ii. Upon satisfaction in full of all Conditions Precedent for a Party, the other Party shall forthwith issue to such Party a Certificate of Compliance with Conditions Precedent (the "Certificate of Compliance").
- iii. The later of the date of issue of Certificate of Compliance to the Concessionaire or the Concessioning Authority shall be the Compliance Date, whereupon the obligations of the Parties under this Concession shall commence and whereon the Concessioning Authority shall issue the **"Notice to Commence"** to the Concessionaire.
- iv. Each Party shall bear its respective costs and expenses of satisfying such Conditions Precedents unless otherwise expressly provided.

2.4.2 Non-fulfillment of Conditions Precedent

- (i) In the event that any of the Conditions Precedents relating to the Concessionaire have not been fulfilled within 90 (Ninety) days of the signing of this Agreement and also, the Concessioning Authority has not waived them fully or partially, this Agreement shall cease to have any effect as of that date and shall be deemed to have been terminated by the mutual agreement of the parties and no Party shall subsequently have any rights or obligations under this Agreement and Concessioning Authority shall not be liable in any manner whatsoever to the Concessionaire or persons claiming through or under it.
- (ii) In the event that the Concessionaire has fulfilled its Conditions Precedent and Concessioning Authority has not procured fulfillment of any or all of the Conditions Precedent set forth in Article 2.4 (b) within the period specified in respect thereof, the Concessioning Authority shall pay to the Concessionaire damages equivalent to an amount calculated at the rate of 0.1% (Zero point one per cent) of the Performance Security for each day's delay until the fulfillment of the Condition Precedent, subject to a maximum of 10% (ten percent) of the Performance Security. In the event when the maximum damages as above has become payable and the Concessioning Authority has still not been able to procure fulfillment of any or all the Conditions Precedent set forth in Article 2.4 (b) and the period for achievement of the same has not been mutually extended then the Concessioning Authority shall be liable to return the Concessionaire the Performance Security. In the event the Concessioning Authority has terminated this Agreement under Article 2.4.2 (i) due to non-fulfillment of Conditions Precedent by the Concessionaire, the Concessioning Authority shall not be liable in any manner whatsoever to the Concessionaire or its contractors, agents and employees and the Concessioning Authority shall forfeit the Performance Security of the Concessionaire
- (iii) In the event that Vacant Possession of the Project Site has been delivered to the Concessionaire prior to the fulfillment in full of the Conditions Precedent, upon the termination of this Agreement the Project Site shall immediately revert to the Concessioning Authority, free and clear from any Encumbrances, irrespective of any outstanding claims between the Parties or any other claims, disputes etc. whatsoever between the Parties.
- (iv) Instead of terminating this Agreement as provided in paragraph (a) above or as the case may be, the Parties may extend the time for fulfilling the Conditions Precedent by mutual agreement.

2.5 Rights and Use of the Project Site

- (i) Pursuant to Article 2.3, the Concessionaire shall have the right to enter upon, occupy and use the Project Site and to make at its costs, charges and expenses such investigations and development activities (including but not limited to land filling, levelling, clearing, shifting of utilities, landscaping and related works including overcoming site constraints, if any) and any other activity as may be necessary or appropriate to implement the Project. However, it is being clarified here that in the event of Termination of this Agreement due to non-fulfilment of Conditions Precedent, Concessioning Authority shall have no liability to make any payment for the activities undertaken by the Concessionaire in the Project Site.
- (ii) Same and except as expressly provided under this Agreement, the Concessionaire shall not mortgage, transfer, assign or otherwise Encumber the Project Site throughout the Concession Period.
- (iii) Subject to the terms of this Agreement, the Concessionaire shall have the right to develop, create, obtain, set up, construct as the context admits or requires, and operate and maintain the Project by itself or through its Contractors.

- (iv)The Concessionaire shall have the right to use the Project Site in accordance with the provisions of this Agreement and for this purpose the Concessionaire may regulate the entry and use of the Project Site by the Third Parties.
- (v) The Concessionaire shall not without the prior written approval of Concessioning Authority use the Project Site for any purpose other than for the purpose of the Project and purposes incidental or ancillary thereto.
- (vi)The Concessionaire has the option to explore possibility of installing solar panel or setting up small generator sets for electricity for each floating cottage.
- (vii) The Concessionaire has the right to re-develop or set-up new cottages.

2.6 Peaceful Possession

The Concessioning Authority hereby warrants that the Project Site together with the necessary right of way has been allotted by the government through the due process of law; and

2.7 "Concession Period" or "Term" of Concession Agreement

a. This Agreement shall be valid and binding on the Parties throughout the period commencing from the Compliance date till the end of 30 (Thirty) years and the Concessionaire hereby accepts the Concession and agrees to implement the Project subject to and in accordance with the terms and conditions of this Agreement.

Provided that in the event of early Termination, the period of Agreement shall be limited to the period commencing from the Compliance Date and ending with the Termination Date.

- b. Upon expiry of the Concession Period as per Article 2.7, the Concessionaire shall peacefully surrender and yield the Project Site and the Project Facility along with Transfer Assets to the Concessioning Authority in accordance with the provisions of Article 7.6 (Hand back of Transfer Assets).
- 2.8 Project Milestone Not applicable

2.8.1 Project Milestone Completion Certificate Not applicable

2.9 Consideration to Concessioning Authority

2.9.1 Annual Concession Fee

- a. In consideration of the rights, privileges and interests granted by the Concessioning Authority to the Concessionaire in terms of this Agreement, the Concessionaire shall pay to the Concessioning Authority or its nominee / designated recipient an Annual Concession Fee, escalated by 5% for every following year.
- b. The first Annual Concession Fee shall be due and payable from the date of completion of 12th month, starting from the Compliance Date and accordingly, the Concessionaire shall deposit the Annual Concession Fee 7 (seven) days prior to the same date every year, throughout the Concession Period.
- c. The Annual Concession Fee shall be payable by the Concessionaire to the Concessioning Authority recipient at the Concessioning Authority's or its nominee's / designated recipient's bank account at a scheduled bank (to be intimated later on) payable at Dehradun.
- In the event of delay up to one week in payment by the Concessionaire, the Concessionaire shall be required to pay the Concessioning Authority or its nominee / designated recipient interest at the rate of SBI PLR plus 4% per annum. In the case of delay beyond such four weeks, it shall be a Concessionaire Event of Default.
- **2.9.2** In consideration of enjoying the right to use the Project Site, the Concessionaire shall pay a License Fee of Rs 1/- (Rs One Only) annually for the duration of the Concession Period as per the terms of the License Agreement.

ARTICLE 3 OBLIGATIONS OF CONCESSIONAIRE

3. In addition to and not in derogation or substitution of any of its other obligations under this Agreement, the Concessionaire shall have the following obligations:

3.1 **Project Development**

Not applicable.

3.2 Performance Security

- a. For due and punctual performance of its obligations under this Agreement, relating to the Project, the Concessionaire shall deliver to Concessioning Authority and maintain Performance Security for an amount not less than equivalent to twelve (12) months' *pro rata* Annual Concession Feein the form of a Bank Guarantee in favor of Chief Executive Officer, Uttarakhand Tourism Development Board; and the Bank Guarantee shall be prepared and submitted as per the format given in Schedule 5 of this Agreement.
- b. The Performance Security shall be valid for a period of 12 (twelve) months and shall be renewed every year, at least 30 (thirty) days prior to the date of expiry of the Bank Guarantee. All charges, fees, costs and expenses related to the Bank Guarantee shall be borne and paid by the Concessionaire. The Concessionaire undertakes and warrants to Concessioning Authority that bank guarantee furnished as above shall be unconditional and irrevocable and shall continue to be effective and enforceable 6 (six) months beyond the date of the expiry of the Concession Period.
- c. Upon occurrence of Concessionaire Event of Default, the Concessioning Authority shall, without prejudice to its other rights and remedies hereunder or in law, be entitled to encash and appropriate relevant amounts from the Performance Security as damages. Upon such encashment and appropriation from the Performance Security, the Concessionaire shall, within 30 (thirty) days replenish the Performance Security, in case of partial appropriation, to its original level before such encashment or appropriation of and in case of appropriation of entire Performance Security to provide a fresh Performance Security and the Concessionaire shall, within the time so granted replenish or furnish to the Concessioning Authority a fresh Performance Security as aforesaid, failing which the Concessioning Authority shall be entitled to terminate this Agreement.
- d. The Performance Security shall be valid till the end of the Concession Period and transfer of the Project along with the Project Site to Concessioning Authority. If the Agreement is terminated due to any event other than a Concessionaire Event of Default, the Performance Security shall, subject to the Concessioning Authority's right to receive or recover amounts, if any, due from the Concessionaire under this Agreement, be duly discharged and released to the Concessionaire.

3.3 Financing Arrangement, Payment of Taxes, etc.

- a) The Concessionaire shall at its cost, expenses and risk make such financing arrangement as would be necessary to develop and implement the Project and to meet all of its obligations under this Agreement, within a period of 180 days from the Appointed Date.
- b) The Concessionaire shall pay all duties, taxes, levies, etc. such Import Duties, etc. towards all or any of the equipment, material, furnishings, etc. to be installed in the Project apart from taking any necessary approvals / permissions for the same.
- c) The Concessionaire shall pay at its own cost all applicable, existing and future taxes / charges / fees / levies including electric bill any property tax, house tax, luxury tax, GST, stamp duty, registration charges and / or pay cost of any legal documentation charges, etc. in respect of the said Project / Project Site, as leviable.

3.4 Project Implementation

Not applicable.

3.5 Operation and Maintenance

- a) The Concessionaire shall operate and maintain the Project Facility in accordance Good Industry Practice and Applicable Laws. The Concessionaire shall throughout the continuance of this Agreement, have full control and management of Project.
- b) Prior to start the new construction activity in the project, the Concessionaire would maintain and operate the existing infrastructures of the project which are handed over by the Concessioning Authority as per "O&M Manual" Schedule 4.
- c) Prior to the start of operations in new construction activities, the Concessionaire would submit to the Concessioning Authority, an Operations and Maintenance Manual for the Project Schedule - 4 ("O&M Manual") setting out in detail the standard operating procedures, schedules, periodicity and other details of the operation and maintenance activities to be carried out for the Project.
- d) The Concessionaire may undertake operations and maintenance of the Project Facility by itself or through a Contractor possessing requisite technical, financial and managerial expertise / capability, provided Concessionaire shall be solely responsible for implementation and operation maintenance of the Project in accordance with the provisions of this Agreement and it shall not relieve the Concessionaire of its obligations and liabilities under this Agreement in any manner.
- e) The Concessionaire shall with due diligence carry out all necessary and periodical tests in accordance with the applicable regulations and Applicable Laws. The Concessionaire shall maintain proper record of such Tests and the remedial measures taken to cure the defects or deficiencies, if any, indicated by the Test results.
- f) The Concessionaire shall be entitled at its own costs, expenses and consequences to perform the following:
 - i. Institution and supervision of operating policies, principles, systems and procedures for all departments including purchasing, accounting, credit management and maintenance, personnel, etc.
 - ii. Instituting, conducting, defending, compromising, referring to arbitration and abandoning any legal or other proceedings, claims and disputes in which the said Project is concerned.
 - iii. To issue its receipts and invoice in its own name.
- g) The Concessionaire will take professional indemnity insurance in respect of the negligence of its Project Facility staff members, employees, executives, Authorities in the operation and management of the Project. The Concessionaire shall throughout the term of this Agreement maintain the professional indemnity insurance and from time to time regularly bear and pay the insurance premium in this regard.
- h) The Concessionaire shall bear and pay all expenses, taxes, due and payable in respect of the operation and management of the Project, including sales tax, GST where applicable, luxury tax and the like and will ensure that these are paid in time and no defaults are made in respect of the same. Further the Concessionaire shall ensure that all returns in this regard are filed before the due date as per Applicable Laws.
- i) The Concessionaire shall not in the operation of the Project do any act or omission which shall be unlawful in nature and throughout the term abide

by all the law, rules, orders, regulations and other requirements as applicable for the operation and management of the Project from time to time.

- j) The Concessionaire shall keep and maintain all such books, records and reports as are required to be maintained under Applicable Laws and shall submit or cause to be submitted to appropriate authorities, all information and reports as are legally required.
- k) The Concessionaire will furnish to the Concessioning Authority such information as required by the Concessioning Authority with respect to operation and management of the Project.
- The Concessionaire shall ensure safe waste disposal from the Facility. All sewerage sump; solid waste shall be disposed of safely as per the relevant guidelines of the State Government. The Concessionaire shall also ensure compliance of all relevant Rules and Regulations for water, air and sound pollution, as may be notified by the State Government from time to time.
- m) The Concessionaire should ensure that adequate focus is given to saving life and property at the Facility. Suitable signage for warning the visitors of the suitable hazards should be prominently displayed at the Facility in English and Hindi. The Concessionaire should also make suitable announcements in English, Hindi and Local language over a Public Addressable system for any like instructions to the visitors.
- n) The Operator shall ensure that no illegal or immoral or such activities take place within the site. The Concessionaire shall be solely responsible for ensuring that the environment of the place is not vitiated by such illegal and immoral acts of some people.
- o) Notwithstanding anything contained herein, the Concessionaire shall be solely responsible for all the claims or proceedings filed with respect to the management and operation of the Project whether with regard to any negligence committed in the Project or by reason of deficiency in the service at the Project or otherwise for any reason whatsoever whether intentionally or unintentionally and the Concessioning Authority shall not be responsible or liable for the same whether vicariously or otherwise and the Concessionaire shall always keep the Concessioning Authority indemnified in this regard.
- p) The Concessionaire will ensure that the highest service standards and good industry practices will be adopted and followed in the said Project throughout the Concession Period.
- q) In the event the Concessionaire has failed to operate and maintain the Project in accordance with the Good Industry Practices and as per the provisions of this Agreement, a notice to that effect will be issued by Concessioning Authority to the Concessionaire. The Concessionaire shall take all steps to address the concerns raised by Concessioning Authority in the said notice within reasonable period. If the Concessionaire does not remedy any such breach in the period specified in the notice, it shall be constituted as "Concessionaire Event of Default" in terms of the provisions of Article 7.
- r) The Concessionaire may subcontract operations/ operation and maintenance of parts of the Project Facility, provided however, that it shall be obligatory upon the Concessionaire to provide copies of all such Project Agreements entered into from time to time with Third Parties to the Concessioning Authority, with incorporated Clauses, in all Project Agreements with such respective Third Parties, that:- (i) the period of the Project Agreement shall be co-terminus with the Term of this Concession Agreement and (ii) in case of Termination of this Concession Agreement (by efflux of time or premature), Concessioning Authority shall have the option of continuing with such Project Agreements such that all proceeds / rentals, arising out from such Project Agreements of the Project Facility, in case they continue to

accrue, shall henceforth be accrued / appropriated to the Concessioning Authority.

- s) The Concessionaire shall have the right to permit/ allow and charge for advertisement/ hoarding or other commercial activity in the premises as per applicable Laws, provided necessary approval is taken from UTDB.
- t) The Concessionaire shall take all necessary clearances and precautionary measures to avoid any accidents.
- u) The Concessionaire may add rooms/ cottage in existing facility with prior approval of the Authority.

3.6 Minimum Equity Requirement (Concessionaire Shareholding)

- a) The shareholding of Lead Member of the Successful Bidder as mentioned in Recital E i.e. ______, in the issued and paid up equity share capital of the Concessionaire shall be not less than:
 - (i) 51% (fifty one percent) during entire agreement period and;
- b) All other members, except O&M Member, of the Consortium shall legally and beneficially hold not less than 11% (eleven percent) of the paid-up equity capital of the Concessionaire during entire agreement period.
- c) At no stage shall any change in the Equity Components / shareholding patterns be made by the Consortium Members or by any of the Associates without obtaining prior approval from Concessioning Authority. On an application made for the purpose, Concessioning Authority may permit the change of equity components / shareholding patterns, provided that the Concessioning Authority is satisfied that the proposed changes shall be in the interest of the implementation of the Project in future and would not be detrimental to any of the rights or interests of the Concessioning Authority. However, no such change in the equity components / shareholding pattern shall be permitted by Concessioning Authority, which would make the Consortium members non-compliant with Articles 3.6(a) and 3.6(b) above.
- d) In case any such change in composition of Consortium has been agreed upon, the modified Consortium would be required to submit a revised Memorandum of Understanding to the Concessioning Authority. In the event of non-compliance of Article 3.6 (a) and 3.6 (b) above, the same shall be construed as Concessionaire Event of Default, and the Concessioning Authority shall be entitled to terminate this Agreement in accordance with Article 7.

3.7 Insurance

- a) At all times during the period of this Agreement, Concessionaire shall at its cost and expense, purchase and maintain by due re-instatement or otherwise, all insurances in respect of the Project Site and Project Facility in accordance with the Good Industry Practice, as required for existing infrastructure handed over by the Concessioning Authority and required under the Financing Documents ("**Insurance Cover**").
- b) All insurance policies in respect of the insurance obtained by the Concessionaire pursuant to this Article 3.7 (a) shall include a waiver of any and all rights of subrogation or recovery of the insurers there under against, *inter alia*, the Concessioning Authority, and its assigns, successors, undertakings and their subsidiaries, affiliates, employees, insurers and underwriters, and of any right of the insurers to any set-off or counterclaim or any other deduction, whether by attachment or otherwise, in respect of any liability of any such person insured under any such policy or in any way connected with any loss, liability or obligation covered by such policies of insurance.

3.8 General Obligations

The Concessionaire shall:

a. procure and maintain in full force and effect, as necessary, appropriate proprietary rights, licenses, agreements and permissions for materials, methods, processes and systems used in or incorporated into the Project;

- b. at all times, to afford access to the Project Site to the authorized representatives of Concessioning Authority, Senior Lenders, other persons duly authorized by any Governmental Agency having jurisdiction over the Project, to inspect the project and to investigate any matter within their authority and upon reasonable notice.
- c. Perform and fulfill its obligations under the Financing Documents.
- d. make efforts to maintain harmony and good industrial relations among the personnel employed in connection with the performance of its obligations under this Agreement and shall be solely responsible for compliance with all labour laws and solely liable for all possible claims and employment related liabilities of its staff employed in relation with the Project and hereby indemnifies Concessioning Authority against any claims, damages, expenses or losses in this regard and that in no case and shall for no purpose shall Concessioning Authority be treated as employer in this regard;
- e. make its own arrangements for construction materials and observe and fulfil the environmental and other requirements under the Applicable Laws and Applicable Permits from time to time;
- f. ensure that the Project Site remains free from all encroachments and take all steps necessary to remove encroachments, if any;
- g. pay all taxes, duties, levies and outgoings, including utility charges relating to the Project;
- h. not do or omit to do any act, deed or thing which may in any manner be violative of any of the provisions of this Agreement; and
- i. Transfer the Project to the Concessioning Authority in fully operational and functional condition upon termination of this Agreement, in accordance with the provisions thereof.
- j. Ensure that the quality of treated affluent solid waste, emissions etc., conform to the standards laid down by the Competent Authorities.

3.9 No Breach of Obligations

The Concessionaire shall not be considered to be in breach of its obligations under this Agreement nor shall it incur or suffer any liability if and to the extent performance of any of its obligations under this Agreement is affected by or on account of any of the following:

- a. Force Majeure Event, subject to Article 6.3;
- b. Concessioning Authority Event of Default;
- c. Compliance with the directions of any Government Agency, other than instructions issued as a consequence of a breach by the Concessionaire of any of its obligations hereunder; and
- d. Closure of the Project or part thereof with the approval of the Concessioning Authority.

ARTICLE 4 OBLIGATIONS OF CONCESSIONING AUTHORITY

- **4.1** In addition to and not in derogation or substitution of any of its other obligations under this Agreement, Concessioning Authority shall have the following obligations:
 - a. Any claims, liabilities and litigations arising out of in providing the Project Site free of Encumbrances, prior to the date of issue of Notice to Commence, shall be borne solely by the Concessioning Authority and it shall indemnify the Concessionaire and shall hold it harmless from any such liabilities that may arise as a result of any such transfer of the Project Site.
 - b. To facilitate the Concessionaire in getting all such approvals, permissions and authorisations which the Concessionaire may require or is obliged to seek under this Agreement in connection with development of the Project and the performance of its obligations hereunder;
 - c. Upon written request from the Concessionaire, facilitate the Concessionaire in obtaining necessary infrastructure facilities and utilities, including water, electricity and telecommunication facilities at rates and on terms no less favourable to the Concessionaire than those generally available to commercial customers receiving substantially equivalent facilities / utilities; and
 - d. The Concessioning Authority shall deliver possession of the Project Site within 60 days from the Appointed Date. Failure to do so, for any reason other than Concessionaire Event of Default, shall constitute an event of default by Concessioning Authority ("Concessioning Authority Event of Default").
 - e. Upon written request from the Concessionaire, provide reasonable facilitation to the Concessionaire for the purpose of discharging its obligations under this Agreement by the Concessionaire
 - f. The Concessioning Authority, through itself, or its nominee may undertake the overall monitoring of the Project Facility during the construction and operation, maintenance & construction period, to verify the structural safety aspects of the Project Facility throughout the Concession Period.

ARTICLE 5 **DELETED**

ARTICLE 6 FORCE MAJEURE

6.1 Force Majeure Event and Change in Law

6.1.1 Any of the following events which is beyond the control of the Party claiming to be affected thereby ("Affected Party") and which the Affected Party has been unable to overcome or prevent despite exercise of due care and diligence, and prevents the Affected Party from performing or discharging its obligations under this Agreement, shall constitute Force Majeure Event

- a) earthquake, flood, inundation and landslide
- b) storm, tempest, hurricane, cyclone, lightning, thunder or other extreme atmospheric disturbances
- c) fire caused by reasons not attributable to the Concessionaire or any of the employees, Contractors or agents appointed by the Concessionaire for purposes of the Project;
- d) acts of terrorism;
- e) strikes, labour disruptions or any other industrial disturbances not arising on account of the acts or omissions of the Concessionaire or the Contractor;
- f) action of a Government Agency having Material Adverse Effect including but not limited to
 - i. acts of expropriation, compulsory acquisition or takeover by any Government Agency of the Project Facility or any part thereof or of the Concessionaire's or the Contractor's rights in Contractor's rights under any of the Project Agreements.
 - ii. any judgment or order of a court of competent jurisdiction or statutory authority in India made against the Concessionaire or the Contractor in any proceedings which is non-collusive and duly prosecuted by the Concessionaire, and
 - iii. any unlawful, unauthorised or without jurisdiction refusal to issue or to renew or the revocation of any Applicable Permits, in each case, for reasons other than Concessionaire's or the Contractor's breach or failure in complying with the Project Requirements, Applicable Laws, Applicable Permits, any judgment or order of a Governmental Agency or of any contract by which the Concessionaire or the Contractor as the case may be is bound.
 - iv. early determination of this Agreement by UTDB for reasons of national emergency, national security or the national interest.
 - v. any failure or delay of a Contractor caused by any of the events mentioned in (f) and above, for which no offsetting compensation is payable to the Concessionaire by or on behalf of the Contractor.
 - vi. war, hostilities (whether declared or not), invasion, act of foreign enemy, rebellion, riots, weapon conflict or military actions, civil war, ionising radiation, contamination by radioactivity from nuclear fuel, any nuclear waste, radioactive toxic explosion, volcanic eruptions, any failure or delay of a Contractor caused by the events mentioned in this sub-clause for which no offsetting compensation is payable to the Concessionaire by or on behalf of the Contractor.

6.2 Obligations of the Parties

- a) As soon as practicable and in any case within seven (7) days of the date of occurrence of a Force Majeure Event or the date of knowledge thereof, the Affected Party shall notify the other Party of the same setting out, *inter alia*, the following in reasonable detail:
 - (i) the nature and extent of the Force Majeure Event;
 - (ii) the estimated duration of the Force Majeure Event;
 - (iii) the nature of and the extent to which, performance of any of its obligations under this Agreement is affected by the Force Majeure Event;
 - (iv) the measures which the Affected Party has taken or proposes to take to alleviate/mitigate the impact of the Force Majeure Event and to resume performance of such of its obligations affected thereby; and

- (v) any other relevant information concerning the Force Majeure Event, and /or the rights and obligations of the Parties under this Agreement.
- As soon as practicable and in any case within five (5) days of notification by the Affected Party in accordance with the preceding Clause 6.2 (a), the Parties shall meet and hold discussions in good faith and where necessary conduct physical inspection/survey of the Project Facility in order to:
 - (i) assess the impact of the underlying Force Majeure Event,
 - (ii) to determine the likely duration of Force Majeure Event and,
 - (iii) to formulate damage mitigation measures and steps to be undertaken by the Parties for resumption of obligations, the performance of which shall have been affected by the underlying Force Majeure Event.
- The Affected Party shall during the duration of Force Majeure event provide to the other Party with regular (not less than fortnightly) reports concerning the matters set out in the preceding clause (b) as also any information, details or document, which the Parties may reasonably require.**Performance of Obligations**

If the Affected Party is rendered wholly or partially unable to perform any of its obligations under this Agreement because of a Force Majeure Event, it shall be excused from performance of such obligations to the extent it is unable to perform the same on account of such Force Majeure Event provided that:

- a) due notice of the Force Majeure Event has been given as required by the preceding Clause 6.2;
- b) the excuse from performance shall be of no greater scope and of no longer duration than is necessitated by the Force Majeure Event;
- c) the Affected Party has taken all reasonable efforts to avoid, prevent, mitigate and limit damage, if any, caused or is likely to be caused to the Project Facility as a result of the Force Majeure Event and to restore the Project Facility, in accordance with the Good Industry Practice and its relative obligations under this Agreement;
- d) when the Affected Party is able to resume performance of its obligations under this Agreement, it shall give to the other Party written notice to that effect and shall promptly resume performance of its obligations hereunder, the non-issue of such notice being no excuse for any delay for resuming such performance;
- e) the Affected Party shall continue to perform such of its obligations which are not affected by the Force Majeure Event and which are capable of being performed in accordance with this Agreement;
- f) any insurance proceeds received shall, subject to the provisions of Financing Documents, be entirely applied to repair, replace or re-instate the assets damaged on account of the Force Majeure Event, or in accordance with Good Industry Practice.

6.4 Termination due to Force Majeure Event

(a) Termination

- (i) If a Force Majeure Event, is an event described under Clauses 6.1.1(a) to (e) continues or is in the reasonable judgement of the Parties likely to continue beyond a period of 120 days, the Parties may mutually decide to terminate this Agreement or continue this Agreement on mutually agreed revised terms. If the Parties are unable to reach an agreement in this regard, the Affected Party shall after the expiry of the said period of 120 days, be entitled to terminate this Agreement.
- (ii) If the Force Majeure Event is an event described in 6.1.1 (f) and the Concessionaire having exhausted the remedies available to him under the Applicable Laws, has been unable to secure the remedy, the Concessionaire shall be entitled to terminate this Agreement.

b)

c)

Provided that the Parties may by mutual agreement, decide to continue this Agreement on revised terms or to terminate this Agreement, if the event described in 6.1.1 (f) subsists or is likely to subsist for a period exceeding 180 days, then either Party shall be entitled to terminate this Agreement.

Provided further, UTDB may at its sole discretion have the option to terminate this Agreement any time after the occurrence of any event described under Clauses 6.1.1(f).

(b) Termination Notice

If either Party, having become entitled to do so, decides to terminate this Agreement pursuant to the preceding clause 6.4(a) (i) or (a) (ii), it shall issue Termination Notice setting out;

- (i) in sufficient detail the underlying Force Majeure Event;
- (ii) the Termination Date which shall be a date occurring not earlier than 60 days from the date of Termination Notice;
- (iii) the estimated Termination Payment including the details of computation thereof and;
- (iv) any other relevant information.

(c) **Obligation of Parties**

Following issue of Termination Notice by either Party, the Parties shall promptly take all such steps as may be necessary or required to ensure that;

- (i) the Termination Payment, if any, payable by UTDB in accordance with the following clause is paid to the Concessionaire on the Termination Date and
- (ii) the Project Facility are handed back to UTDB by the Concessionaire on the Termination Date free from all Encumbrance.

(d) Termination Payment

Upon Termination of this Agreement due to a Force Majeure Event, Termination Payment shall be made to the Concessionaire by UTDB in accordance with the following:

- (i) If Termination is due to a Force Majeure Event, described under Clauses 6.1.1(a) to (e), no Termination Payment shall be made by UTDB to the Concessionaire but, the Concessionaire shall be entitled to receive and appropriate the proceeds of any amounts under insurance policies.
- (ii) If Termination is due to the occurrence of any event described under Clause 6.1.1(f), UTDB shall pay to the Concessionaire Termination Payment equal to 75% of the Book Value

Provided UTDB shall be entitled to deduct from the Termination Payment any amount due and recoverable by UTDB from the Concessionaire as on the Termination Date.

(iii) If Termination is due to the occurrence of any event described under Clause 6.1.1, UTDB shall, pay to the Concessionaire, Termination payment equal to 70% of the Book Value.

Provided UTDB shall be entitled to deduct from the Termination Payment any amount due and recoverable by UTDB from the Concessionaire as on the Termination Date.

6.5 Liability for other losses, damages etc.

Save and except as expressly provided in this Article, neither Party hereto shall be liable in any manner whatsoever to the other Party in respect of any loss, damage, cost, expense, claims, demands and proceedings relating to or arising out of occurrence or existence of any Force Majeure Event.

6.6 Change in Law

- (a) Change in Law shall mean the occurrence or coming into force of any of the following, after the Appointed Date:
 - (i) the enactment of any new Indian law;
 - (ii) the repeal, modification or re-enactment of any existing Indian law;
 - (iii) a change in the interpretation or application of any Indian law by a court of record.

Provided that Change in Law shall not include:

- (i) coming into effect, after the Appointed Date, of any provision or statute which is already in place as of the Appointed Date,
- (ii) any new law or any change in the existing law under the active consideration of or in the contemplation of any government as of the Appointed Date which is a matter of public knowledge,
- (iii) any change in the rates of the Central Taxes.
- (b) Subject to Change in Law resulting in Material Adverse Effect and subject to the Concessionaire taking necessary measures to mitigate the impact or likely impact of Change in Law on the Project, if as a consequence of a Change in Law, the Concessionaire is obliged to incur Additional Costs, UTDB may consider subsequently reimburse to the Concessionaire, but not obliged to.
- (c) Upon occurrence of a Change in Law, the Concessionaire may, notify UTDB of the following:
 - (i) the nature and the impact of Change in Law on the Project
 - (ii) in sufficient detail, the estimate of the Additional Cost likely to be incurred by the Concessionaire on account of Change in Law
 - (iii) the measures, which the Concessionaire has taken or proposes to take to mitigate the impact of Change in Law, including in particular, minimising the Additional Cost
 - (iv) the relief sought by the Concessionaire
- (d) Upon receipt of the notice of Change in Law issued by the Concessionaire pursuant to preceding sub-clause, UTDB and the Concessionaire along shall hold discussions and take all such steps as may be necessary as to determine the quantum of relief to be provided by UTDB to the Concessionaire.
- (e) UTDB shall within 30 days from the date of determination of quantum of Additional Cost, provide relief to the Concessionaire in the manner as mutually agreed upon by the Parties. Escalation, if any, in Additional Cost after such date of determination shall not be payable by UTDB.

ARTICLE 7 EVENTS OF DEFAULT AND TERMINATION

7.1 Events of Default

Events of Default shall mean either Concessionaire Event of Default or Concessioning Authority Event of Default or both as the context may admit or requires.

a. Concessionaire Event of Default

Any of the following events shall constitute an Event of Default by the Concessionaire ("Concessionaire Event of Default") unless such event has occurred as a result of one or more reasons set out in Article 6.1.1:-

- The Concessionaire has failed to achieve Project Milestone and obtain Project Milestone Completion Certificate, within the time period laid down in Article 2.8;
- (ii) The Concessionaire has failed to pay the Annual Concession Fee.
- (iii) The Concessionaire has failed to comply with the Applicable Laws / Applicable Permits or any Rules and such failure has resulted in a Material Breach of the Agreement;
- (iv) The Concessionaire has failed to implement the Project in accordance with the provisions of this Agreement and such failure, in the reasonable estimation of Concessioning Authority, is likely to failure in operation, maintenance and achieving Project Construction Completion Date.
- (v) The Concessionaire's failure to perform or discharge any of its obligations under this Agreement, which has or is likely to have a Material Adverse Effect;
- (vi) The Concessionaire has failed to make any payments due to Concessioning Authority and more than four weeks have elapsed since such payment became due;
- (vii) The Concessionaire is in Material Breach of any of its obligations under this Agreement and the same has not been remedied for more than 90 days;
- (viii) Any representation made or warranty as given by the Concessionaire under this Agreement is found to be false or misleading;
- (ix) A resolution has been passed by the shareholders of the Concessionaire for voluntary winding up of the Concessionaire;
- (x) Any petition for winding up of the Concessionaire has been admitted and liquidator or provisional liquidator has been appointed or the Concessionaire has been ordered to be wound up by court of competent jurisdiction, except for the purpose of amalgamation or reconstruction with the prior consent of Concessioning Authority, provided that, as part of such amalgamation or reconstruction and the amalgamated or reconstructed entity has unconditionally assumed all surviving obligations of the Concessionaire under this Agreement;
- (xi) A default has occurred under any of the Financing Documents and any of the lenders to the Project has recalled its Financial Assistance and demanded payment of the amounts outstanding under the Financing Documents or any of them as applicable;
- (xii) The Concessionaire has abandoned the Project for the consecutive period of 3 (three) months;
- (xiii) The Concessionaire has unlawfully repudiated this Agreement or has otherwise expressed an intention not to be bound by this Agreement;
- (xiv) The Concessionaire has suffered an attachment levied on any of its assets, which has caused or is likely to cause a Material Adverse Affect on the Project and such attachment has continued for a period exceeding 120 days.
- (xv) The Concessionaire has created encumbrance(s) beyond the Concession Period / Term of this Concession Agreement and in violation of the provisions herein in this regard.

b. Concessioning Authority Event of Default

Failure of Concessioning Authority to deliver possession of the Project Site within 60 (Sixty) days of execution of this Agreement for any reason other than Concessionaire Event of Default shall constitute an event of default by Concessioning Authority ("Concessioning Authority Event of Default").

7.2 Termination due to Event of Default

a. Termination for Concessionaire Event of Default

- (i) Without prejudice to any other right or remedy which Concessioning Authority may have in respect thereof under this Agreement, upon the occurrence of a Concessionaire Event of Default, Concessioning Authority shall be entitled to terminate this Agreement in the manner as set out under Article 7.2(a)(ii) and Article 7.2(a)(iii).
- (ii) If Concessioning Authority decides to terminate this Agreement pursuant to preceding (i), it shall in the first instance issue "Preliminary Notice to Remedy" to the Concessionaire and within 30 (Thirty) days of receipt of the Preliminary Notice to Remedy, the Concessionaire shall submit to Concessioning Authority in sufficient detail, the manner in which it proposes to cure the underlying Event of Default (the "Concessionaire's Proposal to Rectify"). In case of non-submission of the Concessioning Authority shall be entitled to terminate this Agreement by issuing the Termination Notice.
- (iii) If the Concessionaire's Proposal to Rectify is submitted within the period stipulated therefore, the Concessionaire shall have further period of 60 (Sixty) days ("Cure Period") to remedy / cure the underlying Event of Default. If, however the Concessionaire fails to remedy / cure the underlying Event of Default within such further period allowed, Concessioning Authority shall be entitled to terminate this Agreement by issuing the Termination Notice.

b. Termination for Concessioning Authority Event of Default

- (i) Without prejudice to any other right or remedy which the Concessionaire may have in respect thereof under this Agreement, upon the occurrence of Concessioning Authority Event of Default, the Concessionaire shall be entitled to terminate this Agreement by issuing Termination Notice.
- (ii) If the Concessionaire decides to terminate this Agreement pursuant to preceding the above Clause (i), it shall in the first instance issue Preliminary Notice to Remedy to the Concessioning Authority. Within 30 (Thirty) days of receipt of Preliminary Notice to Remedy, to Concessioning Authority shall forward to the Concessionaire its proposal to remedy / cure the underlying Event of Default (the "Concessioning Authority Proposal to Rectify"). In case of nonsubmission of Concessioning Authority Proposal to Rectify within the said period of 30 (Thirty) days, Concessionaire shall be entitled to terminate this Agreement by issuing Termination Notice.
- (iii) If Concessioning Authority Proposal to Rectify is forwarded to the Concessionaire within the period stipulated thereof, Concessioning Authority shall have further period of 60 (Sixty) days to remedy / cure the underlying Event of Default. If, however Concessioning Authority fails to remedy / cure the underlying Event of Default within such further period allowed, the Concessionaire shall be entitled to terminate this Agreement by issuing Termination Notice.

c. Termination Notice

If a Party having become entitled to do so decides to terminate this Agreement pursuant to the preceding Sub article 7.2 (a) or 7.2(b), it shall issue Termination Notice setting out:

- (i) in sufficient detail the underlying Event of Default;
- (ii) the Termination Date which shall be a date occurring not earlier than 60 days from the date of Termination Notice;
- (iii) the estimated Termination Payment including the details of computation thereof; and,
- (iv) Any other relevant information.

d. **Obligation of Parties**

Following issue of Termination Notice by either Party, the Parties shall promptly take all such steps as may be necessary or required to ensure that:

- (i) until Termination the Parties shall, to the fullest extent possible, discharge their respective obligations so as to maintain the continued operation of the Project Facility;
- (ii) The Project Facility and Project Site shall be transferred to the Concessioning Authority by the Concessionaire on the Termination Date, free from all Encumbrances, as per the provisions of Clause 7.6 (Hand back of Transfer Assets) of this Agreement.
- (iii) The Termination Payment, if any, payable by either Party in accordance with the following sub clause (f) is paid to the Other Party on the Termination Date.

e. Withdrawal of Termination Notice

Notwithstanding anything inconsistent contained in this Agreement, if the Party who has been served with the Termination Notice cures the underlying Event of Default to the satisfaction of the other Party at any time before the Termination occurs, the Termination Notice shall be withdrawn by the Party which had issued the same.

Provided that the Party in breach shall compensate the other Party for any direct costs / consequences occasioned by the Event of Default, which caused the issue of Termination Notice.

f. Termination Payments

- 1. Upon Termination of this Agreement on account of Concessionaire Event of Default, prior to the Project Construction Completion Date, Concessioning Authority shall retain payments, if any, already received from the Concessionaire.
- 2. Upon Termination of this Agreement on account of Concessionaire Event of Default, the Concessioning Authority shall not make any payments to the Concessionaire. In addition, Concessioning Authority shall also be entitled to forfeit and retain the Performance Security, if subsisting.
- 3. Upon Termination of this Agreement on account of Concessioning Authority Event of Default, Concessioning Authority shall be liable to pay compensation which shall be the higher of following but not more than amount equivalent to concession fee for 3 months:
 - a) The aggregate of net value, as determined by an independent expert, of (i) tangible assets forming part of, fixed or attached to the ground created, installed or provided by the Concessionaire and comprised in the Project, which in the reasonable judgement of the said expert are capable of being

put to use / utilised by the CONCESSIONAIRE and (ii) the moveable assets which the CONCESSIONAIRE agrees to take over LESS any amount due to CONCESSIONING AUTHORITY from Concessionaire under this Agreement and the insurance claims claimed or received. AND

- b) Debt Dues
- 4. All payments due to the Concessionaire as calculated under Article 7.2 (f) shall be made within 30 (thirty) days of receiving the Termination Notice.

7.3 Rights of Concessioning Authority on Termination

Upon Termination of this Agreement for any reason whatsoever, Concessioning Authority shall upon making the Termination Payment, if any, to the Concessionaire have the power and authority to:

- a. Enter upon and take possession and control of the Project Facility and Project Site forthwith as per the provisions of Clause 7.6 (Hand back of Transfer Assets) of this Agreement.
- b. prohibit the Concessionaire and any person claiming through or under the Concessionaire from entering upon / dealing with the Project Facility and Project Site;
- c. Notwithstanding anything contained in this Agreement, Concessioning Authority shall not, as a consequence of Termination or otherwise, have any obligation whatsoever including but not limited to obligations as to compensation for loss of employment, continuance or regularisation of employment, absorption or re-employment on any ground, in relation to any person in the employment of or engaged by the Concessionaire in connection with the Project, and the transfer of the Project Site and Project Facility by the Concessionaire to Concessioning Authority shall be free from any such obligation.

7.4 Accrued Rights of Parties

Notwithstanding anything to the contrary contained in this Agreement, Termination pursuant to any of the provisions of this Agreement shall be without prejudice to accrued rights of either Party including its right to claim and recover monetary damages and other rights and remedies which it may have under law or contract. The rights and obligations of either Party under this Agreement, including without limitation those relating to the Termination Payment, shall survive the Termination but only to the extent such survival is necessary for giving effect to such rights and obligations.

7.5 Expiry of "Term" of Concession Agreement

At the end of the Term of Concession Agreement as per Article 2.7, the Concessioning Authority would either run the Project Facility itself or offer it for private sector participation.

7.6 Hand back of Transfer Assets

Upon Termination of this Agreement for any reason whatsoever or the Expiry of Term of Concession Agreement as mentioned under Article 7.5 and consequent obligation of Concessionaire hand over **"Transfer Assets"** to the Concessioning Authority, on expiry of the Term, the Concessionaire shall ensure that on the Transfer Date its interest in:

- (i) All Movable Assets, movable property, stocks, materials, vehicles and spares relatable to Transfer Assets shall be transferred to Concessioning Authority or its nominee, clear of any Encumbrances and with good title;
- (ii) All assets, duly certified by the concerned regulatory / licensing / accreditation / rating agency, as given below:
 - a. Structural Safety Certificate of the Project Facility from competent, reputed and recognized Structural Engineer
 - b. Shall be transferred to Concessioning Authority or its nomine, clear of Encumbrances and with good title;
- (iii) All immovable property, assets, structures, buildings, services, furnishings, edifices, court-areas, ways, walls, compounds relatable to Transfer Assets shall be transferred to Concessioning Authority or its nominee, clear of all Encumbrances and with good title;
- (iv) The rights and obligations under or pursuant to all contracts relatable to Transfer Assets and other arrangements entered into in accordance with the provisions of this Agreement between Concessionaire and any third party shall (in consideration of Concessioning Authority's assumption of the obligations under or pursuant to the contracts and other arrangements) be vested in Concessioning Authority or its nominee, clear of all Encumbrances and with good title; and
- (v) Notwithstanding anything contained in Article 7.6 (i),(ii), (iii) and (iv), prior to any transfer of the Transfer Assets, Concessioning Authority, shall have the right to conduct a due diligence of the contracts and agreements, the rights and obligations of which it is assuming and shall not be bound to assume the rights and obligations of contracts that, in the sole opinion of Concessioning Authority are unreasonably onerous, and would be considered onerous at the time that the contracts were entered into. In relation to all such contracts that are not transferred to Concessioning Authority, no third Entity, including the counterparty of such contract shall have any right, license, title, interest, benefit, claim or demand against or over any Asset and such Transfer Asset shall be transferred to Concessioning Authority or its nominee, free from all Encumbrances and with good title.
- (vi) Furthermore, notwithstanding anything contained in the Article 11.1 no liability (accrued or contingent) of Concessionaire or relating to the Project Facility arising on account of actions or inactions prior to the Transfer Date shall be assumed or transferred to Concessioning Authority or its nominees. The Concessioning Authority or its nominees shall not be liable for liabilities in relation to the Project arising pursuant to the Transfer Date. In the event of any such liability being assumed or transferred to Concessioning Authority or its nominee or any Encumbrance existing on any of the Transfer Assets, the quantum of such liability and / or amount corresponding to such Encumbrance, shall be compensated by Concessionaire to Concessioning Authority.
- (vii) Without prejudice to the foregoing, Concessionaire agrees to indemnify and keep indemnified the Concessioning Authority from and against all actions, proceedings, losses, damages, liabilities, claims, costs and expenses whatsoever which may be sustained or suffered by the Concessioning Authority as a result of any actions or omissions of the Concessionaire prior to the transfer of the Transfer Assets. It is expressly understood by the Parties that this Article shall survive the termination or expiry of this Agreement.
- (viii) Vesting Certificate: Hand back of Transfer Assets is deemed to be complete on the date when all the requirements under Article 6.6 have been fulfilled by the Concessionaire, and the Concessioning Authority shall, without unreasonable delay, thereupon issue a certificate substantially in the form set forth in Schedule-6 (the "Vesting Certificate"), which will have the effect of constituting evidence of divestment by the Concessionaire of all of its rights, title and interest in the Project, and their vesting in the Concessioning Authority pursuant hereto. It is expressly agreed that the issue of Vesting Certificate shall

not in any manner be construed or interpreted as restricting the exercise of any rights by the Concessioning Authority or its nominee on, or in respect of, the Project on the footing that all requirements under this Agreement have been complied with by the Concessionaire.

ARTICLE 8 DISPUTE RESOLUTION

8.1 Amicable Resolution

- a. Save where expressly stated to the contrary in this Agreement, any dispute, difference or controversy of whatever nature between the Parties, howsoever arising under, out of or in relation to this Agreement (the "**Dispute**") shall in the first instance be attempted to be resolved amicably in accordance with the procedure set forth in Sub-article 8.1 (b) below.
- b. Either Party may require such Dispute to be referred to the CEO, UTDB and Chairman of Board of Directors of the Concessionaire, for amicable settlement. Upon such reference, the two shall meet at the earliest mutual convenience and in any case within 30 (Thirty) days of such reference to discuss and attempt to amicably resolve the Dispute, as evidenced by the signing of written terms of settlement within 30 (Thirty) days of such meeting or such longer period as may be mutually agreed by the Parties. If the Dispute is not amicably settled within 30 (Thirty) days (or such longer mutually agreed period) of such meeting between the two, either Party may refer the Dispute to arbitration in accordance with the provisions of Article 8.2 below.

8.2 Arbitration

a. **Procedure**

Subject to the provisions of Article 8.1, any dispute, which is not resolved amicably, shall be finally decided by reference to arbitration under the Arbitration Act. The arbitration shall be by a panel of three arbitrators, one to be appointed by each Party and the third to be appointed by the two arbitrators appointed by the Parties. The Party requiring arbitration shall appoint an arbitrator in writing, inform the other Party about such appointment and call upon the other Party to appoint its arbitrator. If within 15 days of receipt of such intimation the other Party fails to appoint its arbitrator, the Party seeking appointment of arbitrator may take further steps in accordance with the Arbitration Act.

b. Place of Arbitration

The place of arbitration shall be Dehradun only and the jurisdiction of the Courts of Dehradun shall prevail.

c. English Language

The request for arbitration, the answer to the request, the terms of reference, any written submissions, any orders and awards shall be in English and, if oral hearings take place, English shall be the language to be used in the hearings.

d. Enforcement of Award

The Parties agree that the decision or award resulting from arbitration shall be final and binding upon the Parties and shall be enforceable in accordance with the provisions of the Arbitration Act subject to the rights of the aggrieved parties to secure relief from any higher forum.

e. Performance during Arbitration

Pending the submission of and / or decision on a dispute and until the arbitral award is published; the Parties shall continue to perform their respective obligations under this Agreement, without prejudice to a final adjustment in accordance with such award.

8.3 Adjudication by Regulatory Authority

In the event of the constitution of a statutory Regulatory Authority with powers to adjudicate upon Disputes between the Concessionaire and the Concessioning Authority, then instead of reference to Arbitration under Article 8.2, such Regulatory Authority

shall adjudicate upon all Disputes arising out after its constitution, in accordance with the Applicable Laws.

ARTICLE 9 FINANCING AND LENDER'S RIGHTS

- **9.1** The Concessioning Authority hereby agrees that it shall enter into a Substitution Agreement, with the Senior Lender and the Concessionaire thereby shall grant the Senior Lender certain rights in the event of "Termination for Concessionaire Event of Default" as set out in Article 6.2 (a). The Concessioning Authority further confirms to the Concessionaire that it can represent to the Senior Lenders that the Concessioning Authority has agreed to be bound by the terms and conditions specified therein.
- 9.2 At any time after the Lenders' Representative (defined under Substitution Agreement) have issued a Notice of Financial Default of the Concessionaire. it may by notice require the Concessioning Authority to suspend all the rights of the Concessionaire and undertake the operation and maintenance of the Project on its own or through a Third Party. Upon receipt of such notice, the Concessioning Authority shall undertake the operation of the Project during such "Suspension" period. The aforesaid Suspension shall be revoked upon substitution of the Concessionaire by a Nominated Company (defined under Substitution Agreement), and in the event such substitution is not completed within 180 (one hundred and eighty) days from the date of such Suspension, the Concessioning Authority may terminate the Concession Agreement forthwith by issuing a Termination Notice in accordance with the provisions of this Concession Agreement; provided that upon written request from the Lenders' Representative and the Concessionaire, the Concessioning Authority may extend the aforesaid period of 180 (one hundred and eighty) days by a period not exceeding 90 (ninety) days.

ARTICLE 10 REPRESENTATIONS AND WARRANTIES

10.1 Representations and Warranties of the Concessionaire

The Concessionaire represents and warrants to Concessioning Authority that:

- a. it is duly organised, validly existing and in good standing under the laws of India;
- b. it has full power and authority to execute, deliver and perform its obligations under this Agreement and to carry out the transactions contemplated hereby;
- c. it has taken all necessary corporate and other action under Applicable Laws and its constitutional documents to authorise the execution, delivery and performance of this Agreement;
- d. it has the financial standing and capacity to undertake the Project;
- e. this Agreement constitutes its legal, valid and binding obligation enforceable against it in accordance with the terms hereof;
- f. the information furnished in the bid / proposal and as updated on or before the date of this Agreement is true and accurate in all respects as on the date of this Agreement;
- g. the execution, delivery and performance of this Agreement will not conflict with, result in the breach of, constitute a default under or accelerate performance required by any of the terms of the Concessionaire's Memorandum and Articles of Association or any Applicable Laws or any covenant, agreement, understanding, decree or order to which it is a party or by which it or any of its properties or assets are bound or affected;
- h. there are no actions, suits, proceedings or investigations pending or to the Concessionaire's knowledge threatened against it at law or in equity before any court or before any other judicial, quasi-judicial or other authority, the UTDB of which may constitute Concessionaire Event of Default or which individually or in the aggregate may result in Material Adverse Effect;
- i. it has no knowledge of any violation or default with respect to any order, writ, injunction or any decree of any court or any legally binding order of any Government Agency which may result in Material Adverse Effect;
- j. it has complied with all Applicable Laws and has not been subject to any fines, penalties, injunctive relief or any other civil or criminal liabilities which in the aggregate have or may have Material Adverse Effect;
- k. subject to receipt by the Concessionaire from Concessioning Authority of any amount due under any of the provisions of this Agreement, in the manner and to the extent provided for under the applicable provisions of this Agreement all rights and interests of the Concessionaire in and to the Project Site and Project Facility shall pass to and vest in Concessioning Authority on the Transfer Date free and clear of all Encumbrances without any further act or deed on the part of the Concessionaire or Concessioning Authority;
- 1. no representation or warranty by the Concessionaire contained herein or in any other document furnished by it to Concessioning Authority or to any Government Agency in relation to Applicable Permits contains or will contain any untrue statement of material fact or omits or will omit to state a material fact necessary to make such representation or warranty not misleading; and
- m. Without prejudice to any express provision contained in this Agreement, the Concessionaire acknowledges that prior to the execution of this Agreement, the Concessionaire has after a complete and careful examination made an independent evaluation of the Project Site, and the information provided by Concessioning Authority, and has determined to its satisfaction the nature and extent of risks and hazards as are likely to arise or may be faced by the Concessionaire in the course of performance of its obligations hereunder.

The Concessionaire also acknowledges and hereby accepts the risk of inadequacy, mistake or error in or relating to any of the matters set forth above and hereby confirms

that Concessioning Authority shall not be liable for the same in any manner whatsoever to the Concessionaire

10.2 Representations and Warranties of Concessioning Authority

Concessioning Authority represents and warrants to the Concessionaire that:

- a. Concessioning Authority has full power and authority to enter into this Agreement and has taken all necessary action to authorise the execution, delivery and performance of this Agreement;
- b. This Agreement constitutes Concessioning Authority's legal, valid and binding obligation enforceable against it in accordance with the terms hereof;
- c. It has a good and valid right over the Project Site and has power and authority to grant Concession in respect thereto to the Concessionaire; and
- d. Upon the Concessionaire paying the Annual Concession Fee and performing the covenants herein, it shall not at any time during the Term hereof, interfere with peaceful exercise of the rights and discharge of its obligations by the Concessionaire, in accordance with this Agreement.

10.3 Obligation to Notify Change

In the event that any of the representations or warranties made / given by a Party ceases to be true or stands changed, the Party who had made such representation or given such warranty shall promptly notify the other of the same.

ARTICLE 11 MISCELLANEOUS

11.1 Assignment and Charges

The Concessionaire shall not assign in favour of any person this Agreement or the rights, benefits and obligations hereunder, save and except with prior consent of Concessioning Authority.

11.2 Interest and Right of Set Off

Any sum which becomes payable under any of the provisions of this Agreement by one Party to the other Party shall, if the same be not paid within the time allowed for payment thereof, shall be deemed to be a debt owed by the Party responsible for payment thereof to the Party entitled to receive the same. Such sum shall until payment thereof carry interest @ SBI PLR as on 31st March 2017 may be replaced to just "SBI PLR prevailing from time to time", from the due date for payment thereof until the same is paid to or otherwise realised by the Party entitled to the same. Without prejudice to any other right or remedy that may be available under this Agreement or otherwise under law, the Party entitled to receive such amount shall also have the right of set off.

Provided the stipulation regarding interest for delayed payments contained in this Article 11.2 shall neither be deemed nor construed to authorise any delay in payment of any amount due by a Party nor be deemed or construed to be a waiver of the underlying breach of payment obligations.

11.3 Governing Law and Jurisdiction

This Agreement shall be governed by the laws of India. The Courts of Dehradun only shall have jurisdiction over all matters arising out of or relating to this Agreement.

11.4 Waiver

- a. Waiver by either Party of any default by the other Party in the observance and performance of any provision of or obligations under this Agreement:
 - (i) shall not operate or be construed as a waiver of any other or subsequent default hereof or of other provisions or obligations under this Agreement;
 - (ii) shall not be effective unless it is in writing and executed by a duly authorised representative of such Party; and
 - (iii) Shall not affect the validity or enforceability of this Agreement in any manner.
- b. Neither the failure by either Party to insist on any occasion upon the performance of the terms, conditions and provisions of this Agreement or any obligation hereunder nor time or other indulgence granted by a Party to the other Party shall be treated or deemed as waiver / breach of any terms, conditions or provisions of this Agreement.

11.5 Survival

Termination of this Agreement:

- a. shall not relieve the Concessionaire or Concessioning Authority of any obligations already incurred hereunder which expressly or by implication survives Termination hereof; and
- b. except as otherwise provided in any provision of this Agreement expressly limiting the liability of either Party, shall not relieve either Party of any obligations or liabilities for loss or damage to the other Party arising out of or caused by acts or omissions of such Party, prior to the effectiveness of such Termination or arising out of such Termination.

11.6 Amendments

This Agreement and the Schedules together constitute a complete and exclusive understanding of the terms of the Agreement between the Parties on the subject hereof and no amendment or modification hereto shall be valid and effective unless agreed to by both the Parties hereto and evidenced in writing.

11.7 Notices

Unless otherwise stated, notices to be given under this Agreement including but not limited to a notice of waiver of any term, breach of any term of this Agreement and termination of this Agreement, shall be in writing and shall be given by hand delivery, recognised international courier, mail, telex or facsimile transmission and delivered or transmitted to the Parties at their respective addresses set forth below:

If to Concessioning Authority:

Chief Executive Officer Uttarakhand Tourism Development Board, Pt. Deen Dayal Upadhyay, Paryatan Bhawan, Near O.N.G.C. Helipad, Nimbuwala, Garhi Cantt, Dehradun – 248 003

Email: ceo.tourism.uk@gmail.com

If to the CONCESSIONAIRE:

Fax No.

Or such address, telex number, or facsimile number as may be duly notified by the respective Parties from time to time, and shall be deemed to have been made or delivered.

- a. in the case of any communication made by letter, when delivered by hand, by recognised international courier or by mail (registered, return receipt requested) at that address; and
- b. In the case of any communication made by telex or facsimile, when transmitted properly addressed to such telex number or facsimile number.

11.8 Severability

If for any reason whatsoever any provision of this Agreement is or becomes invalid, illegal or unenforceable or is declared by any court of competent jurisdiction or any other instrumentality to be invalid, illegal or unenforceable, the validity, legality or enforceability of the remaining provisions shall not be affected in any manner, and the Parties shall negotiate in good faith with a view to agreeing upon one or more provisions which may be substituted for such invalid, unenforceable or illegal provisions, as nearly as is practicable. Provided failure to agree upon any such provisions shall not be subject to dispute resolution under this Agreement or otherwise.

11.9 No Partnership

Nothing contained in this Agreement shall be construed or interpreted as constituting a partnership between the Parties. Neither Party shall have any authority to bind the other in any manner whatsoever.

11.10 Language

All notices required to be given under this Agreement and all communications, documentation and proceedings which are in any way relevant to this Agreement shall be in writing and in English language.

11.11 Exclusion of Implied Warranties etc.

This Agreement expressly excludes any warranty, condition or other undertaking implied at law or by custom or otherwise arising out of any other agreement between the Parties and any representation by any Party not contained in a binding legal agreement executed by the Parties.

11.12 Counterparts

This Agreement may be executed in 2 counterparts, each of which when executed and delivered shall constitute an original of this Agreement but shall together constitute one and only the Agreement.

11.13 General Liability and Indemnity

- a. The Concessionaire shall indemnify, defend and hold Concessioning Authority harmless against any and all proceedings, actions and Third Party claims arising out of a breach by Concessionaire and its contractors, sub-contractors, agents, employees, etc. of any of its obligations under this Agreement except to the extent that any such claim has arisen due to breach by Concessioning Authority of any of its obligations under this Agreement.
- b. The Concessioning Authority shall, indemnify, defend and hold harmless the Concessionaire against any and all proceedings, actions, Third Party claims for loss, damage and expense of whatever kind and nature arising out of breach by Concessioning Authority, its officers, servants and agents of any obligations of Concessioning Authority under this Agreement except to the extent that any such claim has arisen due to breach by the Concessionaire of any of its obligations under this Agreement.

IN WITNESS WHEREOF the Parties have executed this Agreement as of the day and year first above written.

For and on behalf of Concessioning Authority Signed by

For and on behalf of Concessionaire Signed by

In the presence of Witnesses: 1) 2)

SCHEDULE 1 PROJECT SITE INCLUDING DETAILS OF EXISTING INFRASTRUCTURE

Land details		Other details
Approximately 11 acre of land	1.	Visitor facilitation/ Information centre and an
		Interpretation centre
	2.	Convention centre
	3.	Cottage -16 Numbers
	4.	Executive Bedroom – 4 Numbers
	5.	Deluxe Bedroom – 4 Numbers
	6.	Overhead water storage tank
	7.	Electrical room
	8.	Staff and driver dormitory
	9.	Manager room
	10.	Parking area
	11.	Musical & Dancing Fountain
	12.	Food court
	13.	Generator shed
	14.	Artificial Climbing Wall

SCHEDULE 2 CONCESSION PAYMENT

Not Applicable

SCHEDULE 3 SCOPE OF CONSTRUCTION & MINIMUM CONSTRUCTION REQUIREMENT

As submitted by concessionaire & approved by UTDB. To be finalized at the time of signing of Contract

SCHEDULE 4 OPERATION AND MAINTENANCE REQUIREMENTS

To be finalized at the time of signing of Contract

SCHEDULE 5 FORMAT OF PERFORMANCE SECURITY (BANK GUARANTEE)¹

PERFORMANCE BANK GUARANTEE OF CONCESSIONAIRE

Bank Guarantee No.:

Dated:

Issuer of Bank Guarantee:

(Name of the Bank)

(Hereinafter referred to as the **"Bank"**)

Beneficiary of Bank Guarantee:

UTDB

Nature of Bank Guarantee:

Unconditional and irrevocable Bank Guarantee.

Context of Bank Guarantee

Performance in respect of Concession Agreement (hereinafter referred to as the "Agreement") to be entered between the Uttarakhand Tourism Development Board (hereinafter referred to as the "Concessioning Authority") and _______ (hereinafter referred to as the "Concessionaire") for the Operation, Maintenance ofon Public Private Partnership (PPP) basis at, (hereinafter referred to as the "Project"), provided however, such context of the Bank Guarantee or reference to the Agreement in this Bank Guarantee shall in no manner to relied upon at any stage to adversely affect or dilute the unconditional and irrevocable nature of this Bank Guarantee. The title of this Guarantee i.e. "Performance Bank Guarantee" shall in no manner and at no stage be relied upon to adversely affect or dilute the unconditional and irrevocable nature of this Bank Guarantee.

Operative part of the Bank Guarantee:

- 3. We hereby waive the necessity of the Concessioning Authority demanding the said amount from Concessionaire prior to serving the Demand Notice upon us.

¹ To be issued by a Scheduled Bank in India

- 4. We further agree and affirm that no change or addition to or other modification to the terms of the Agreement, shall in any way release us from any liability under this unconditional and irrevocable Guarantee and we hereby waive notice of any such change, addition or modification. We further agree with the Concessioning Authority that the Concessioning Authority shall be the sole and the exclusive judge to determine that whether or not any sum or sums are due and payable to him by Concessionaire, which are recoverable by the Concessioning Authority by invocation of this Guarantee.
- 5. This Guarantee will not be discharged due to the change in constitution of the Bank or the Concessionaire. We undertake not to withdraw or revoke this Guarantee during its currency / validity period, except with the previous written consent of the Concessioning Authority.
- 6. We unconditionally and irrevocably undertake to pay to the Concessioning Authority, any amount so demanded not exceeding Rs. Lakhs (Rupees Lakhs (Rupees Only) notwithstanding any dispute or disputes raised by Concessionaire or anyone else in any suit or proceedings before any dispute review expert, arbitrator, court, tribunal or other authority, our liability under this Guarantee being absolute, unconditional and unequivocal. The payment so made by us under this Guarantee to the Concessioning Authority, shall be a valid discharge of our liability for payment under this Guarantee and the Concessionaire shall have no claim against us for making such payment.
- 7. This unconditional and irrevocable Guarantee shall remain in full force and effect and shall remain valid until ______

Notwithstanding anything contained herein:

- 1. Our liability under this Bank Guarantee shall not exceed Indian Rs.Lakhs (Rupeesonly).
- 2. This unconditional and irrevocable Bank Guarantee shall be valid w.e.f. _____to
- 3. We are liable to pay the guaranteed amount or any part thereof under this unconditional and irrevocable Bank Guarantee only and only if the Concessioning Authority serves upon us a written claim or demand on or before _____.

Authorized Signatory

For Bank

SCHEDULE 6 VESTING CERTIFIEATE

- 1. The UTDB (the "**Concessioning Authority**") refers to the Concession Agreement dated *** (the "**Agreement**") entered into between the Concessioning Authority and **** (the "**Concessionaire**") for Operation, Maintenance ofon Public Private Partnership (PPP) basis.
- 2. The Concessioning Authority hereby acknowledges compliance and fulfilment by the Concessionaire of the requirements set forth in Article 6.6 of the Agreement on the basis that upon issue of this Vesting Certificate, the Concessioning Authority shall be deemed to have acquired, and all title and interest of the Concessionaire in or about the Project shall be deemed to have vested unto the Government, free from any encumbrances, charges and liens whatsoever.
- 3. Notwithstanding anything to the contrary contained hereinabove, it shall be a condition of this Vesting Certificate that nothing contained herein shall be construed or interpreted as waiving the obligation of the Concessionaire to rectify and remedy any defect or deficiency in any of the requirements under the Agreement and/or relieving the Concessionaire in any manner of the same

Signed this *** day of ***, 2021 at [***].

AGREED, ACCEPTED AND SIGNED SIGNED, SEALED AND DELIVERED

For and on behalf of

For and on behalf of

CONCESSIONAIRE by: (Signature) (Name) (Designation)

(Address)

CONCESSIONING AUTHORITY by

(Signature) (Name)

(Designation) (Address)

In the presence of:

1.

2.

SCHEDULE 7 Deleted

SCHEDULE 8

License Agreement

This License Agreement is executed at -----(*insert name of place*), on this the _____ day of_____;

$B \to T W \to E N$

(Name of the Concessioning Authority) (hereinafter referred to as the **"Licensor"** (which expression unless it be repugnant to the context or meaning thereof shall mean and include their respective heirs, legal representatives, administrators, executors and assigns) of the **FIRST PART**

AND

(*Name of the person who is being given the sub-licence*) hereinafter referred to as the "Licensee" (which expression unless it be repugnant to the context or meaning thereof shall mean and include its successors and assigns) of the SECOND PART

Capitalized terms used in this Agreement, but not defined shall have the meaning assigned to them in the Concession Agreement.

WHEREAS

- A. The Licensor is the Concessioning Authority under the Concession Agreement dated ("the Concession Agreement") entered into between Uttarakhand Tourism Development Board, ("UTDB") and the Concessionaire.
- B. The Licensor has been granted, *inter alia*, the right to enter into License Agreements with suitable Persons in respect of the Additional Facility (Name of the Facility......), subject to the terms set forth in the Concession Agreement.
- C. Pursuant to the rights granted to the Licensor as aforesaid, the Licensor has agreed to give on license basis to the Licensee, the Licensed Premises No. ______ more particularly described in Appendix hereto ("the Licensed Premises") for a period of ____years with effect from ______ upto _____, at a License Fee of Rs. _____ per year, for use by the Licensee of the Licensed Premises in accordance with the terms of this Agreement and in accordance with applicable laws;

NOW THEREFORE, THIS AGREEMENT WITNESSETH AS FOLLOWS:

- 1. The Licensor does hereby grant and the Licensee does hereby accept the license to use, occupy and enjoy the Licensed Premises for a term of _____ years commencing from _____ upto _____ ("the License Period"), subject to the terms and conditions hereinafter set out. The term of the License could be renewed/ extended for a further period and on such terms as may be mutually agreed between the parties.
- 2. The Licensee shall pay to the Licensor, a License Fee of Rs. _____ per month ("the License Fee") during the License Period.
- **3**. The License Fee shall be payable monthly in advance, on or before the seventh day of each month.

- 4. The Licensee shall pay to the Licensor an amount equal to ______months License Fee as an Advance i.e. Rs.______ ("Advance"). The Licensor agrees to return the Advance to the Licensee upon termination of this Agreement, after adjusting any amounts that may be due to the Licensor.
- 5. The Licensor shall make available adequate facilities for maintenance of cleaning of the common areas such as corridor, staircase, toilets, elevators, parking area.
- 6. The Licensor shall be responsible for the maintenance of all the common areas in the Additional Facility. All cost related to maintenance as aforesaid, would be borne by the Licensor. The Licensor agrees to promptly rectify all defects in the common areas etc. brought to its notice.
- 7. The Licensee shall not do or suffer to be done anything in the Licensed Premises, which is or is likely to be a nuisance or annoyance to the other occupants of the Additional Facility or to prejudice the rights of the Licensor of the Licensed Premises under the Concession Agreement.
- 8. The Licensee shall take due and proper care of the Licensed Premises and shall be liable to compensate the Licensor for any loss or damage to the Licensed Premises, save and except for reasonable wear and tear. The Licensee shall not carry out any change in the Licensed Premises, without the prior written of the Licensor.
- 9. The Licensee declares, agrees and is aware that this Agreement does not create any right, title to or in the Licensed Premises in favour of the Licensee, save and except a limited right to use, occupy and enjoy the Licensed Premises during the term of this Agreement. Further, the Licensee shall not sub-let the Licensed Premises to any party.
- 10. The Licensee shall be liable to keep the Licensed Premises in good habitable condition and keep the Licensed Premises at his own cost adequately water proofed and carry out such repairs as may be required from time to time and as called upon to do by the Licensor.
- 11. The Licensee does hereby indemnify and undertake to keep indemnified the Licensor from all damages, claims, liabilities arising out of or as a consequence of any action taken or penalty imposed by any Person or authority on account of use of the Licensed Premises by the Licensee under this Agreement, which action may include an attachment, restraint or any other proceedings.
- 12. The Licensor shall pay the applicable taxes, cess, levies, etc., in respect of the Licensed Premises.
- 13. In the event of either party committing breach of any of the terms and conditions of this Agreement and failing to remedy or make good such breach within 30 days of the receipt of notice in writing from the other party, such other party shall be entitled to revoke this Agreement by giving the other party a notice of 15 days in writing, without payment of any compensation whatsoever.

- 14. In the event of not requiring the Licensed Premises, the Licensee shall be entitled to terminate this Agreement by giving three month's notice in writing to the Licensor or three month's Licensee Fee in lieu thereof.
- 15. In an unlikely event of unforeseen circumstances, which may render difficult for the Licensee to occupy the Licensed Premises, the Licensee shall be entitled to terminate the License by giving one-month notice to the Licensor. Similarly, in an unlikely event of unforeseen circumstances, which may render difficult for the Licensor to give or continue on license basis the Licensed Premises, the Licensor shall be entitled to terminate this Agreement by giving one-month notice to the Licensee.
- 16. The Licensee hereby expressly agrees that this Agreement shall automatically and simultaneously terminate upon termination of the Concession Agreement, for any reason whatsoever including default of the Licensor thereunder, and that the Licensee shall not have any rights whatsoever against the Licensor/UTDB, including compensation or termination payment which the Licensor/UTDB may be entitled to receive under the Concession Agreement. Upon termination of the Concession Agreement between UTDB and Concessionaire/Licensor, UTDB shall have right to enter upon and take possession of the property.
- 17. Upon termination of this Agreement for any reason, the Licensee shall remove all objects and persons using the Licensed Premises at the cost of the Licensee and deliver vacant and peaceful possession of the Licensed Premises therein to the Licensor in a good condition, normal wear and tear excepted.
- 18. Unless otherwise stated, notices to be given under this Agreement including but not limited to a notice of waiver of any term, breach of any term of this Agreement and termination of this Agreement, shall be in writing and shall be given by hand delivery, recognised international courier, mail, telex or facsimile transmission and delivered or transmitted to the Parties at their respective addresses set forth below : If to First Party:

The,	
	,
Fax No.	

If to the Second Party

The,			
	 	·····,	
Fax No.	 		

Or such address, telex number, or facsimile number as may be duly notified by the respective Parties from time to time, and shall be deemed to have been made or delivered

• in the case of any communication made by letter, when delivered by hand, by recognised international courier or by mail (registered, return receipt requested) at that address, and

- in the case of any communication made by telex or facsimile, when transmitted properly addressed to such telex number or facsimile number.
- 19. Notwithstanding anything to the contrary contained in this Agreement, Termination pursuant to any of the provisions of this Agreement shall be without prejudice to accrued rights of either Party including its right to claim and recover money damages and other rights and remedies which it may have in law or contract. The rights and obligations of either Party under this Agreement, including without limitation those relating to the Termination Payment, shall survive the Termination but only to the extent such survival is necessary for giving effect to such rights and obligations.

20. Settlement of Disputes:

- 20.1 Save where expressly stated to the contrary in this Agreement, any dispute, difference or controversy of whatever nature between the Parties, howsoever arising under, out of or in relation to this Agreement, shall in the first instance be attempted to be resolved amicably by meetings between the Parties.
- 20.2 In the case of dispute arising upon or in relation to or in connection with the contract between the Client and the Consultant, which has not been settled amicably, any party can refer the dispute for Arbitration under (Indian) Arbitration and Conciliation Act, 1996 and subsequent amendments from time to time. Such disputes shall be referred to an Arbitral Tribunal consisting of 3 (three) arbitrators, one each to be appointed by the Client and the Consultant, the third arbitrator shall be chosen by the two arbitrators so appointed by the parties and shall act as Presiding Arbitrator. In case of failure of the two arbitrators, appointed by the parties to reach a consensus regarding the appointment of the third arbitrator within a period of 30 days from the date of appointment of Indian Council of Arbitration. The Arbitration and Conciliation Act, 1996 and any statutory modification or re-enactment thereof, shall apply to these arbitration proceedings.
- 20.3 Arbitration proceedings shall be held in India at the place indicated in SC and the language of the arbitration proceedings and that of all documents and communications between the parties shall be English.
- 20.4 The decision of the majority of arbitrators shall be final and binding upon both parties. The expenses of the arbitrators as determined by the arbitrators as per the Act and shall be shared equally by the Client and the Consultant. However, the expenses incurred by each party in connection with the preparation and presentation shall be borne by the party itself. All arbitration awards shall be in writing and shall state the reasons for the Award.
- 20.5 Pending the submission of and/or decision on a dispute, the Parties shall continue to perform their respective obligations under this Agreement without prejudice to a final adjustment in accordance with such arbitration award.
- 20.6 This Agreement shall be governed by the laws of India and courts of the District Headquarters, Dehradun will have jurisdiction relating to all matters arising from this Agreement.
- 20.7 This Agreement and the Schedules together constitute a complete and exclusive understanding of the terms of the Agreement between the Parties on the subject hereof and no amendment or modification hereto shall be valid and effective unless agreed to by all the Parties hereto and evidenced in writing.

IN WITNESS WHEREOF the parties have subscribed their respective hands hereunto on the day, month and year first herein above written.

SIGNED AND DELIVERED by the Licensor by the hand of ______ in the presence of

1. _____

2. _____

SIGNED AND DELIVERED by the Licensee by the hand of ______ in the presence of

- 1. _____
- 2. _____

RECEIPT

Received from	an amount of Rs	[Rupees
	_Only] as Advance as mentioned in Clause 4 of the License Agreement.	-

Date: Place: -----(insert name of place)

SCHEDULE 9

DRAFT PROJECT SITE LEASE DEED

This	Lease	made	at		the	 day	of
				Two Thousand		•	

BETWEEN

______ and having its registered office at ______ (hereinafter referred to as the "Lessor", which expression shall, unless the context does not so admit, include its successors and assigns) of the One Part;

AND

______, a Company Incorporated under the Companies Act, 1956 and having its registered office at ______ (hereinafter called "**the Lessee**", which expression shall unless the context does not so admit includes its successor or successors in business and permitted assigns) of the Other Part;

WHEREAS

		(herei	nafter	referre	ed to	as the "Co	ncessi	oning A	uthor	rity"
"	or the	"Lessor")	has	vide	the	Concessio	on A	greemen	nt d	ated
						(hereir	after	"Co	onces	sion
greement")	entered an	nongst					_ (he	reinafter	refe	rred
as the "		_") and M/s					(he	reinafter	r refe	rred
as the	"Concessie	onaire" or	the	"Less	ee")	granted	Conc	ession	to	M/s
		_ to plan,	design	, fina	nce,	construct,	devel	lop, ope	erate	and
aintain a _		a	t			(hereii	nafter re	ferre	d to
the "Proj	ect") on a	Operation,	Main	tenanc	e	of	01	n Public	e Pri	vate
rtnership (l	PPP) basis	and to detern	nine, 1	levy, d	lemai	nd, collect,	retair	n and ap	prop	riate
riffs and F	ees from the	ne Users of	the Pr	oject]	Facili	ity. Entry t	fee pa	rking cl	narge	s as
ed by Cond	cessionaire.									
	greement") as the " as the intain a the "Proj rtnership (I riffs and F	reement") entered am as the " as the "Concession intain a the "Project") on a rtnership (PPP) basis a	"" or the "Lessor") reement") entered amongst as the "") and M/s as the "Concessionaire" or to plan, or intain aa the "Project") on a Operation, rtnership (PPP) basis and to deterr riffs and Fees from the Users of	"" or the "Lessor") has reement") entered amongst as the "") and M/s as the "Concessionaire" or the to plan, design intain aat the "Project") on a Operation, Main rtnership (PPP) basis and to determine, I riffs and Fees from the Users of the Pr	"" or the "Lessor") has vide preement") entered amongst as the "") and M/s as the "Concessionaire" or the "Less to plan, design, fina intain a at the "Project") on a Operation, Maintenance rtnership (PPP) basis and to determine, levy, design riffs and Fees from the Users of the Project	"" or the "Lessor") has vide the preement") entered amongst as the "") and M/s as the "Concessionaire" or the "Lessee") to plan, design, finance, intain a at the "Project") on a Operation, Maintenance rtnership (PPP) basis and to determine, levy, demand riffs and Fees from the Users of the Project Facility	"" or the "Lessor") has vide the Concession (herein preement") entered amongst as the "") and M/s as the "Concessionaire" or the "Lessee") granted to plan, design, finance, construct, intain a at (the "Project") on a Operation, Maintenanceof rtnership (PPP) basis and to determine, levy, demand, collect, riffs and Fees from the Users of the Project Facility. Entry to	"" or the "Lessor") has vide the Concession A (hereinafter greement") entered amongst (hereinafter greement") entered amongst (hereinafter greement") entered amongst (hereinafter greement") and M/s (hereinafter greement") and M/s (hereinafter greement" or the "Lessee") granted Conce greement the "Concessionaire" or the "Lessee") granted Conce greement to plan, design, finance, construct, development to plan, design, finance, construct, development at (hereinafter greement to plan, design, finance, construct, development the "Project") on a Operation, Maintenanceofofof rtnership (PPP) basis and to determine, levy, demand, collect, retain riffs and Fees from the Users of the Project Facility. Entry fee particular set of the Project Facility. Entry fee particular set of the Project Facility. Entry fee particular set of the Project Facility.	"" or the "Lessor") has vide the Concession Agreement (hereinafter "Concessionaire") entered amongst (hereinafter "Concessionaire") and M/s (hereinafter "Concessionaire" or the "Lessee") granted Concession to plan, design, finance, construct, develop, operationain a at (hereinafter references the "Project") on a Operation, Maintenanceofon Public references and to determine, levy, demand, collect, retain and apriffs and Fees from the Users of the Project Facility. Entry fee parking clients and the project facility.	of the Lessor) has vide the Concession Agreement d (hereinafter "Concess preement") entered amongst (hereinafter refe as the "") and M/s (hereinafter refe as the "Concessionaire" or the "Lessee") granted Concession to to plan, design, finance, construct, develop, operate intain a at (hereinafter referre the "Project") on a Operation, Maintenanceofon Public Pri rtnership (PPP) basis and to determine, levy, demand, collect, retain and appropri riffs and Fees from the Users of the Project Facility. Entry fee parking charge

- B. The vesting of the land comprising the Project Site and all rights relating thereto with the Concessionaire by the Concessioning Authority, under a valid and binding lease deed in a form satisfactory to the Concessionaire, is specified in the Concession Agreement as a Condition Precedent to the Concessionaire undertaking its obligations under the Concession Agreement.
- C. In order to satisfy this Condition Precedent of the Concession Agreement, the Lessor hereby vests with the Lessee the land owned by it comprising the Project Site on the terms and conditions and subject to the covenants and stipulations hereinafter contained.

NOW THIS INDENTURE OF LEASE WITNESSETH AS FOLLOWS:

- 1. The capitalized terms contained herein shall, unless the context otherwise requires, have the same meaning as ascribed to them in the Concession Agreement.
- 2. The following terms shall have the meaning as hereunder:
 - (a) **Deed** means this Lease Deed;
 - (b) **Appendix** means any of the appendices, supplements and documents annexed to this Deed;
 - (c) **Concession Agreement** shall mean the Concession Agreement dated ______ entered into between_____ and

M/s _____; and a copy of which is annexed hereto and marked as Appendix-B to form part of this Lease Deed.

- (d) **Demised Premises** means all the lands which are described, delineated and shown in Appendix "A" hereto and which comprise the Project Site and are owned by the Lessor;
- (e) **Parties** shall mean Lessor and Lessee herein;
- (f) **Lessor** means the Concessioning Authority;
- (g) Lessee means the Concessionaire or _____

3. <u>Interpretation</u>

- i. Words denoting the singular shall include the plural and vice-versa;
- ii. Words denoting a person shall include an individual, corporation, company, partnership firm, joint ventures, trust or other entity;
- iii. Headings and bold typeface are only for convenience and shall be ignored for the purposes of interpretation;
- iv. Reference to any legislation or law or to any provision thereof shall include references to any such law as it may, after the date of this Deed, from time to time be amended, supplemented or re-enacted;
- v. References to the word "include" or "including" shall be construed without limitation;
- vi. References to this Deed or to any other agreement or deed or other instrument shall be construed as a reference to such agreement, deed, or other instrument as the same may from time to time be amended, varied, supplemented or novated;
- vii. Reference to any Party to this Deed or any other agreement or deed or other instrument shall include its successors or permitted assigns;
- viii. A reference to a section, Article, paragraph or Appendix is, unless indicated to the contrary, a reference to a section, Article, paragraph or Appendix to this Deed;
- ix. The Appendices to this Deed are a part of this Deed and shall be enforced as if they were expressly set out in the body of this Deed; and
- x. Any agreement, consent, approval, authorization, proposal, notice, communication, information or report required under or pursuant to this Deed from or by any Party shall be valid and effectual only if it is in writing under the hand of the duly authorised representative of such Party in this behalf and not otherwise.
- 4. The Lessor hereby demises the Demised Premises unto the Lessee under the terms of this Deed from the ______ day of ______ 2021 for a period, which shall be co-terminous with the terms of the Concession Agreement and shall be extended or terminated at a prior date to coincide with the Concession Period. The Lessor hereby undertakes that it shall not terminate this Deed, except upon the due and valid termination of the Concession Agreement.

 - 6. The Lessee shall pay unto the Lessor an annual rent of Re 1/- as lease rental, which shall be paid as an advance lease rental in single lump sum payment of Rs. 90/- (Rupees Ninety Only) on or prior to the date upon which this deed is executed.
 - 7. The Lessor hereby vests the Demised Premises with the Lessee along with all Easementary Rights, free from any Encumbrance; provided that the Lessee shall remove

the structures and public utilities at, on, over the Demised Premises at its cost and expense and the Lessor shall render all necessary assistance in this behalf.

- 8. The Lessor hereby vests the Demised Premises with the Lessee under this Deed for the purpose of implementing the Project, including for the purposes of planning, designing, engineering, financing, construction, operation and maintenance of the Project Facility and to develop, construct, operate and maintain all facilities, structures, buildings, works and activities whatsoever as may be authorised under the rights granted to the Lessee in accordance with the terms of the Concession Agreement or otherwise approved by the Lessor.
- 9. The Parties hereby agree that the Lessee shall not mortgage, transfer, assign or otherwise encumber the Demised Premises, in the form of lease, sub-lease, license, right-of-way or otherwise create a security interest thereon in favour of the Lenders for the purpose of enabling Financing of the Project and for securing finances for permitted activities or purposes under the Concession Agreement.
- 10. Intentionally Left Blank
- 11. In the event of termination of the Concession Agreement by efflux of time or otherwise, this Lease Deed shall stand terminated and the Concessionaire shall hand over the Vacant Possession of the Project Site to the Concessioning Authority or its nominated agency. And the Concessionaire, its contractors, sub-contractors, agents, employees and Licensees shall henceforth vacate the Project Site.
- 12. The Lessor hereby covenants with the Lessee as under:
 - (a) That it shall not increase the lease rental due and payable by the Lessee under Section 6 hereof;
 - (b) That it shall not interfere with or impede in any manner or otherwise limit, restrict or impose any conditions or restrictions on the complete, free and full enjoyment and use of the Demised Premises and all rights in relation thereto by the Lessee,
 - (c) It shall not interfere in or impede in any manner or otherwise limit, restrict or impose conditions in relation to: (i) to the development, construction, implementation, operation and maintenance of the Facility; (ii) the implementation of the Project by the Lessee; and (iii) the possession, control and use by the Lessee of the Demised Premises, the Facility constructed thereon and any other facilities developed in the course of implementation of the Concession Agreement or otherwise with the approval of the Lessor;
 - (d) That except the lease rental specified in Section 6 hereof, it shall not levy any fee, rental, tax or any other charge on the Lessee for the lease of the Demised Premises;
 - (e) That it has clear and marketable title to and is in lawful possession of the Demised Premises and that there is no litigation, claim, demand or any proceedings pending before any Competent Authority in respect of the acquisition of or the Lessor's title to or possession of the Demised Premises and the Lessee shall have complete, lawful and uninterrupted Vacant Possession, control and use of the Demised Premises; and
 - (f) That it shall not terminate this Deed, except upon the due and valid termination of the Concession Agreement in accordance with the terms thereof.
- 13. The Lessee hereby covenants with the Lessor as follows:
 - (a) That it shall develop, establish, design, construct, implement and commence Commercial Operations of the Facility as per its obligations under the Concession Agreement and complete the construction of the Facility within the time period specified thereof in the Concession Agreement;

- (b) That it shall operate and maintain the Facility or cause it to be operated and maintained in accordance with the Concession Agreement;
- (c) That it shall observe and perform all terms, covenants, conditions and stipulations of this Deed;
- (d) That it shall inform the Lessor about encroachments on the Demised Premises that are affecting the performance of its rights under the Concession Agreement. The Lessee shall assist the Lessor in removing these encroachments; and
- (e) That it shall pay all the applicable stamp duty and registration charges for the proper and valid execution and registration of this Deed.
- 14. Each Party hereto represents and warrants that:
 - (a) It has full power and authority to execute, deliver and perform its obligations under this Deed and to carry out the transactions contemplated hereby;
 - (b) It has taken all necessary actions to authorise the execution, delivery and performance of this Deed; and
 - (c) This Deed constitutes its legal, valid and binding obligations that shall be enforceable against it in accordance with the terms hereof.
- 15. The Parties hereby acknowledge and specify that this Deed is being executed for the benefit of the Lessee and the Lessee shall have the right to enforce this Deed through specific performance. The Parties further agree that:
 - (a) The failure by a Party to perform its obligations under this Deed and any breach of covenants or undertakings given and provided for in this Deed by the Party shall amount to an Event of Default under the Concession Agreement ("Event of Default").
 - (b) Any dispute, controversy or claim arising out of or in relation to this Deed or the interpretation of any of its provisions shall be settled in accordance with the provisions of Article 7 of the Concession Agreement.
 - (c) All notices under the terms of this Deed shall be sent either by hand, facsimile or courier to the following addresses:

Lessee :

IN WITNESS WHEREOF the parties hereto have affixed their respective signatures to this Lease Deed the day and year first herein above written:

SIGNED	IN THE NAME AND ON BEHALF	SIGNED IN	THE NAME AND ON BEHALF
OF THE _	THROUGH :	OF M/s	THROUGH :

IN THE PRESENCE OF WITNESSES:

1. 2. Place:

Date:

Appendix A: Particulars of land with detailed project site map / plan Appendix B: Signed Concession Agreement